



# Stranding aboy? Heterogeneous transition beliefs and capital investment choices

Cahen-Fourot, Louison; Campiglio, Emanuele; Daumas, Louis; Miess, Michael Gregor; Yardley, Andrew

Published in: Journal of Economic Behavior and Organization

DOI: 10.1016/j.jebo.2023.10.028

Publication date: 2023

Document Version Publisher's PDF, also known as Version of record

Citation for published version (APA):

Cahen-Fourot, L., Campiglio, E., Dáumas, L., Miess, M. G., & Yardley, A. (2023). Stranding ahoy? Heterogeneous transition beliefs and capital investment choices. Journal of Economic Behavior and Organization, 216, 535-567. https://doi.org/10.1016/j.jebo.2023.10.028

# **General rights**

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
  You may not further distribute the material or use it for any profit-making activity or commercial gain.
- · You may freely distribute the URL identifying the publication in the public portal.

## Take down policy

If you believe that this document breaches copyright please contact rucforsk@kb.dk providing details, and we will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect



# Journal of Economic Behavior and Organization

journal homepage: www.elsevier.com/locate/jebo

**Research Paper** 

# Stranding aboy? Heterogeneous transition beliefs and capital investment choices $\stackrel{\star}{\approx}$

Louison Cahen-Fourot <sup>a,b</sup>, Emanuele Campiglio <sup>c,d,e</sup>, Louis Daumas <sup>f,\*</sup>, Michael Gregor Miess <sup>g,b,h,i</sup>, Andrew Yardley <sup>j</sup>

<sup>a</sup> Roskilde University, Denmark

<sup>b</sup> WU Institute for Ecological Economics, Austria

° University of Bologna, Italy

## <sup>d</sup> RFF-CMCC European Institute on Economics and the Environment, Italy

<sup>e</sup> LSE Grantham Research Institute on Climate Change and the Environment, United Kingdom

<sup>f</sup> Centre International de Recherche en Environnement et Développement, France

g KAUST - King Abdullah University of Science and Technology, Kingdom of Saudi Arabia

h Federal Environment Agency Austria, Austria

<sup>i</sup> Institute for Advanced Studies, Austria

<sup>j</sup> Offshore Renewable Energy Catapult, United Kingdom

# ARTICLE INFO

JEL classification: D84 E71 Q43 Q56 *Keywords:* Asset stranding

Asset stranding Beliefs Heterogeneous expectations Investment choices Low-carbon transition Short-termism

# ABSTRACT

Individuals have heterogeneous beliefs regarding the future speed and shape of the low-carbon transition. In this paper, we study to what extent opinion diversity matters for aggregate capital investment decisions. We develop a model where firms formulate heterogeneous expectations around a dominant narrative, or 'market norm', with their dispersion increasing over a finite planning horizon. Our analytical and numerical results suggest that belief heterogeneity can significantly affect the share of low-carbon investments, with the strength of its effects non-linearly correlated to market norms. We show that investment behaviour tends to be more sensitive to shocks to short-term, rather than long-term, belief heterogeneity, highlighting the importance of setting credible short-term targets. Finally, we find beliefs to interact strongly and in non-trivial ways with measures of short-termism, with increasing agents' farsightedness not necessarily leading to less carbon-intensive investments under high heterogeneity.

# 1. Introduction

How do we expect the low-carbon transition to unroll? This question is likely to be answered in radically different ways by different individuals. The answer depends on their information set, their degree of trust in policy-makers, their beliefs on future

\* Corresponding author.

E-mail address: louis.daumas@enpc.fr (L. Daumas).

https://doi.org/10.1016/j.jebo.2023.10.028

Received 2 December 2022; Received in revised form 7 September 2023; Accepted 22 October 2023

Available online 7 November 2023





<sup>\*</sup> The replication package for the paper is available at https://github.com/SMOOTH-ERC/stranding\_ahoy. We would like to thank Giorgos Galanis, Francesco Lamperti, Hubert Massoni, Esther Marie Shears, Alessandro Spiganti, Marko Stojanovic and Roberta Terranova, as well as participants to the 2021 EAERE and the 2022 ESEE conferences. The research leading to these results has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 Research and Innovation Programme (Grant agreement No 853050 - SMOOTH). EC, LCF and MM also gratefully acknowledge the Oesterreichische Nationalbank (Jubiläumsfonds projects 17641 and 18651).

<sup>0167-2681/© 2023</sup> The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC license (http://creativecommons.org/licenses/by-nc/4.0/).

technological advancements, etc. Some might expect the transition to take place rapidly and in line with limiting global warming to 1.5 °C or 2 °C (IEA, 2021; IPCC, 2022). Others might expect a more gradual dynamics, either allowing carbon-intensive assets to exit the economy in an orderly fashion, or to remain in smaller proportion. Others might not even have well-formulated answers, as the transition timeline goes beyond the span of their planning horizon.

Disagreement over what lies in the future is common and found to be significant in several spheres of economic behaviour (Coibion et al., 2018; Hommes, 2021; Mankiw et al., 2003; Xiong and Yan, 2010). Similarly, some recent contributions have shown evidence of heterogeneous expectations for what concerns transition-related beliefs (*e.g.* Giglio et al., 2023; Nordeng et al., 2021). However, these insights have yet to be incorporated in suitable modelling frameworks. Does expectation heterogeneity matter for the low-carbon transition dynamics?

In this paper, we address this research question by developing a model of capital investment choices with heterogeneous beliefs.<sup>1</sup> Firms choose how to allocate investments between a high- and a low-carbon technology based on their relative expected profits, discounted over a finite planning horizon. Profit expectations are affected by how decision-makers perceive the future speed and shape of the low-carbon transition. We proxy transition-related beliefs by focusing on the expected degree of 'stranding' of the high-carbon technology, *i.e.* its expected utilisation rate. If the firm imagines a rapid low-carbon transition in the near future, it will expect the high-carbon technology to be used less than fully, and hence generate lower revenues and profits. If it expects instead a slow transition, it will anticipate full or close to full utilisation of carbon-intensive capital stocks. We then use a discrete choice model to aggregate the multiple decentralised individual investment choices and compute the overall share of low-carbon investments.

We introduce three main features of expectations in the attempt of capturing relevant real-world dimensions. First, a central transition projection exists, representing the dominant wisdom, or 'market norm', within the community of investors. Agents take this as a focal point when formulating their expectations and investment decisions (Beckert and Bronk, 2018; Schelling, 1960). This 'common wisdom' can take the form of a general narrative that most agents believe to be true and that guides their investment decisions, which might be more or less close to the government's stated policy objectives (Boyer, 2018; Campiglio et al., 2023). While we refrain from simulating co-existing conflicting narratives, we explore the implications of two distinct possible central projections: (i) a low-stranding scenario, roughly aligned with current policies and the expectation of a gradual transition; and (ii) a high-stranding scenario, entailing a more rapid transition and the achievement of net-zero emissions by mid-century.

Second, individual agents might drift from this dominant opinion, following their own beliefs. As exemplified by the wide range of possible decarbonisation pathways given by IPCC (2022) and other institutions (*e.g.* IEA, 2020c), the precise pace and shape of the transition is far from being well-established. Different combinations of production technologies, energy efficiency measures and societal changes may leave investors undecided as to what kind of business will be the most adequate for the future. Further uncertainties around the implementation of mitigation policies and technological developments cast doubt on the pace and steadiness of the low-carbon transition and force agents to formulate their own expectations around the market norm (Nemet et al., 2017). Hence, today's assessment of the profit prospects of available technological options is a distribution, rather than a point value, with its shape depending on the strength of transition-related belief diversity. Heterogeneity in beliefs may make aggregate investment decisions uncoordinated, possibly contradictory, ultimately hampering the good course of the transition (Acemoglu and Jensen, 2018; Fais et al., 2016).

Third, the degree of heterogeneity varies with the length of the time horizon considered. As shown by the literature on the term structure of expectations for key macroeconomic variables, diversity in expectations depends crucially on how far away in time agents are projecting (Binder et al., 2022; El Ouadghiri and Uctum, 2020; Patton and Timmermann, 2010). Fig. 1 confirms this evidence by showing how the dispersion of expectations concerning future carbon price levels increases if individuals are asked to provide estimates for periods further in time. We capture this stylised fact by letting the variance of expectation distribution increase along the planning horizon. We propose a novel logistic characterisation of this feature, building on the IPCC 6th Assessment Report decarbonisation scenarios (IPCC, 2022).

We then calibrate our model on 2019 data for the European Union, and derive both analytical and numerical results. We illustrate our findings with a sensitivity analysis based on sensible value ranges of our behavioural parameters. Our main results can be summarised as follows.

First, heterogeneity of transition expectations does matter, as it significantly affects firms' investment allocation decisions. We find the direction and strength of this effect to depend on the underlying market norm and its associated capital stranding dynamics. In general, stronger belief heterogeneity will decrease the share of low-carbon investment in presence of high-stranding central expectations, and increase it when central expectations forecast low stranding. However, we also find this relationship to be strongly non-linear. Market norms centred around expectations of either very low or very high stranding (*i.e.* very slow or very rapid transition) will lock in investment behaviours, while more balanced stranding expectations are exposed to large investment swings if belief dispersion moves.

Second, we show how, for our benchmark calibration and for a sizeable proportion of the parameter constellation we explore, investment behaviours are more sensitive to shocks to short-term, rather than long-term, belief heterogeneity. This situation only reverses if agents are strongly far-sighted. This result suggests that policy-makers should prioritise anchoring expectations for the earliest periods of the transition, even if there is no clear consensus about long-term outcomes. The introduction of credible *short-term* 

<sup>&</sup>lt;sup>1</sup> In what follows, we use the word 'beliefs' interchangeably with 'expectations' to insist on the idea that beliefs about the future are not model-consistent, following Acemoglu and Jensen (2018).



Fig. 1. Distribution of expected carbon price in the EU Emission Trading Scheme for different time horizons. Adapted from Nemet et al. (2017), using data from Nordeng et al. (2021). (For interpretation of the colours in the figure(s), the reader is referred to the web version of this article.)

targets hence appears crucial. In addition, policy-makers risk misinterpreting the drivers of investment allocation changes (*e.g.* an increase in low-carbon investment share might be due to shocks to short-run belief heterogeneity rather than a change in market norms). This warrants a thorough measurement and analysis of transition-related expectations, currently missing.

Third, we find belief heterogeneity to interact strongly with measures of short-termism (high discount rate or short planning horizon).<sup>2</sup> The effect of a higher discount rate on low-carbon investment share is exacerbated (mitigated) by higher heterogeneity if central expectations imply high (low) stranding. Belief heterogeneity can also negatively affect the impact of longer planning horizon, which generally increases low-carbon investment shares, by further widening the range of projections and expanding the diversity of opinions.

Our article contributes to better understanding the role of expectations and time preferences in defining the carbon intensity of capital investment choices. Several contributions rooted in neoclassical economic theory have investigated *optimal* investment decision-making in the context of the low-carbon transition, with or without uncertainty (Cai and Lontzek, 2018; Campiglio et al., 2022; Van den Bremer and Van der Ploeg, 2021; Vogt-Schilb et al., 2018, among others). Expectations in these models are usually homogeneous and model-consistent. Other contributions in the field of behavioural macroeconomics have studied transition dynamics allowing for belief/preference heterogeneity and stronger complexity in individual and systemic behaviours (*e.g* Dunz et al., 2021; Geisendorf, 2016; Knobloch and Mercure, 2016). A particularly relevant stream of work for us is the one studying the process of technological diffusion (Mercure, 2012; Mercure et al., 2016), which adopts a similar modelling approach rooted in discrete choice theory. However, these models tend to be governed by backward-looking (adaptive) expectations, making it difficult to analyse the effect of changes in longer-term thinking of economic agents. We opt for an in-between modelling strategy, able to capture expectations that are both forward-looking and (dynamically) heterogeneous. Our approach is similar in spirit to the literature developing logit/probit models of switching beliefs. This has been usually applied to issues linked to inflation expectations and monetary policy (De Grauwe and Macchiarelli, 2015; Franke and Westerhoff, 2018; Galanis et al., 2022; Hommes, 2021), but features recent applications to climate- and transition-related matters (Cafferata et al., 2021; Campiglio et al., 2022; Dávila-Fernández and Sordi, 2020; Guilmi et al., 2022; Zeppini, 2015).

Our treatment of transition expectations positions us close to the literature studying the 'stranding' of physical or financial assets along a low-carbon transition (Campiglio and van der Ploeg, 2022; Daumas, 2023; van der Ploeg and Rezai, 2020). To our knowledge, this paper is the first to incorporate capital stranding into a model with heterogeneous expectations, with most other contributions assuming homogeneity (see for instance Baldwin et al., 2020; Campiglio et al., 2022; Rozenberg et al., 2020). We also partly connect to the large field on the role of time preferences, discounting and planning horizons, in climate-related economic dynamics (see Groom et al. (2022) for a review). Finally, our joint representation of idiosyncratic beliefs and market norms calls out an emerging literature in economic sociology (Beckert and Bronk, 2018; Bronk, 2009) and macroeconomics (Andre et al., 2021; Barrero, 2022) that has emphasised the role of beliefs and narratives in driving economic outcomes.

The remainder of the article is structured as follows. Section 2 presents our modelling framework. Section 3 explains our calibration strategy. Section 4 discusses some analytical results and illustrates them numerically. Section 5 concludes and discusses future research avenues.

<sup>&</sup>lt;sup>2</sup> Many companies extend the time horizon considered to adopt business decisions, including investments, only up to a few years into the future (Souder et al., 2021; Spiro, 2014). This tendency can be exacerbated by cognitive limitations, biases or norms, creating sets of incentives favouring myopia. While it has been shown that short-termist behaviour can slow down the development of relevant technologies and the pace of decarbonisation (Löffler et al., 2019; Nerini et al., 2017; Souder et al., 2016), it is yet unclear how different time preferences could interact with heterogeneous transition expectations in determining individual and aggregate capital investment choices.

#### 2. The model

The model is populated by a continuum of firms producing electricity e in response to an exogenous level of electricity demand  $e^d$ , growing at a constant rate  $g_E$ . Electricity can be produced by two technologies: i) a stock of capital  $K_L$  running on renewable resources and producing low-carbon electricity, and ii) a stock of capital  $K_H$  running on fossil fuels and producing high-carbon electricity. The subscript  $i \in \{H, L\}$  denotes the technology type. The electricity produced by the two technologies,  $e_i$ ,  $i \in \{H, L\}$ , is identical. However, capital stocks have different productivities  $\xi_H$  and  $\xi_L$ .<sup>3</sup> Capacity utilisation  $u_i = \frac{e_i}{\xi_i K_i} \in [0, 1]$ , defines the extent

to which capital stocks are used. We also define a full capacity utilisation level  $u_i^f$ , which can differ from one.

Firms decide how to allocate their investments in new physical capital between the two available technologies. To do so, they formulate expectations regarding their future profitability by exploring potential future transition scenarios. We define the space in which these calculations take place as 'psychological time' (*s*), which takes the form of a finite planning horizon [[1; *S*]] to distinguish it from chronological time (*t*). In other words, psychological time is simply the mental projection of the future by individuals, extending for a finite number of periods. This representation of a finite planning horizon is how we represent our behavioural understanding of short-termism. This horizon is common to all firms. Expectations by a firm are denoted using the expectations operator  $\mathbb{E}$ . Chronological time *t* does not move in this paper, as we investigate only the impact of expectations about the future on present decision-making. We thus do not use a chronological time subscript *t* for convenience.

#### 2.1. Technological return rates

We assume a continuum of small firms index by  $j \in \mathbb{R}$ . Firms compare the two available technologies  $i \in \{L, H\}$  by calculating their unitary return rate  $r_i$ , *i.e.*, the sum of the discounted stream of expected profits  $\pi_{i,j,s}$  that can be obtained from a unit of technology (in our case, a unit of installed generation capacity). Firms discount profit expectations over this planning horizon with the same discount factor  $\beta = \frac{1}{1+\rho}$ , with  $\rho$  the corporate discount rate our measure of 'rational' short-termism. For firm *j*, it writes:

$$\mathbb{E}_{j}(r_{i}) = \sum_{s=0}^{S} \beta^{s} \mathbb{E}(\pi_{i,j,s}).$$
(1)

Three components determine the unitary profit rate  $\pi_i$ : i) revenues; ii) capital costs; and iii) variable costs.

Revenues come from producing and selling electricity  $e_i$ . Electricity is sold on a wholesale market with a merit order structure. The energy produced by renewable sources usually comes first in the merit order, as these incur lower marginal costs than fossilbased technologies (Figueiredo and da Silva, 2019). It means that low-carbon electricity will be sold first and, assuming it is not enough by itself to satisfy the entire demand (*i.e.*  $e^d > \xi_L K_L$ ), capacity utilisation  $u_L$  will be equal to  $u_L^f$ . The high-carbon capital stock will instead be used to the extent necessary to satisfy demand not already met by low-carbon electricity. That is,

$$u_H = \frac{e^d - e_L}{\xi_H K_H}.$$
(2)

The price of electricity  $p_E$  is also determined on the merit order, as the price offered by the marginal technology producing (the high-carbon one, typically). Hence, per-period revenues stemming from a unit of capital  $K_i$  can be computed as  $p_E u_i \xi_i$ .

Capital costs are incurred when installing a new capital unit. The cost of installing a unit of capital *i* is  $c_i$ . Firms require external finance to perform new investments. We define  $\psi_i$  as the debt-to-investment ratio, *i.e.* the proportion of investment expenditure funded via borrowed credit. Companies must pay back the debt over the course of the loan tenure *LT*, together with accrued interests. Firms spread repayment tranches equally throughout the loan tenure period. We thus calculate a capital recovery factor  $\alpha_i = \frac{i_i(1+i_i)^{LT_i}}{1-(1+i_i)^{LT_i}}$ , where *i* is the fixed interest rate applied on the loan.  $\alpha$  represents the ratio between the period repayments (inclusive of principal and interests) and the loan stock. Per-period capital costs for technology *i* can thus be computed as  $\alpha_i \psi_i c_i$ .

Variable costs in the model only arise from purchasing fossil fuels as intermediate inputs necessary to operate  $K_H$  (we abstract for simplicity from other variable costs such as labour). We name the price of fossil fuels  $p_F$  and the productivity of fossil fuels in producing electricity  $\xi_F$ . Variable costs can thus be computed as  $\frac{p_F \xi_H u_H}{\xi_F}$ .

At time *t*, we can treat most of the parameters above as constant in the expected future.<sup>4</sup> The price of electricity  $p_E$  is more and more determined by long-term power purchase agreements and is, therefore, less subjected to uncertainty over long horizons. Capital productivity parameters  $\xi_i$  are embodied in the specific vintage of capital available today. While productivity is likely to change in the future in more advanced capital vintages, we explicitly place ourselves in the context of an investor comparing current technologies and anticipating their payoffs. Sidestepping from simple productivity losses due to ageing and related maintenance costs, we assume investors to consider productivity parameters to be constant over their planning horizon. Finally, we treat the price of fossil fuels  $p_F$  as exogenous, as this allows us to disentangle the effect of stranding expectations.

<sup>&</sup>lt;sup>3</sup> Since capital is the only production input in our model, productivities  $\xi_H$  and  $\xi_L$  can also be interpreted as Leontief production function coefficients.

<sup>&</sup>lt;sup>4</sup> In Appendix B, we extend the model by allowing for heterogeneity in beliefs on the future profitability of both technologies, which include prices of both electricity and fossil fuels.

The proportion of investments covered by debt  $\psi_i$  is decided today and remains the same. Loan duration *LT* and interest rate *i* are negotiated with the bank today and remain constant. In other words, firms choose a fixed interest rate and do not renegotiate financing conditions.

We can now define the expected unitary retained profit rates<sup>5</sup> for the two technologies as:

$$\mathbb{E}_{j}(\pi_{L,j,s}) = p_{E}\xi_{L}u_{L}^{j} - \alpha_{L}\psi_{L}c_{L},$$

$$\mathbb{E}_{j}(\pi_{H,j,s}) = \left(p_{E} - \frac{p_{F}}{\xi_{E}}\right)\xi_{H}\mathbb{E}_{j}(u_{H,s}) - \alpha_{H}\psi_{H}c_{H},$$
(4)

where the only variable subject to firms' expectations is the capacity utilisation rate of the high-carbon capital stock,  $u_H$ . We denote by  $\mathbb{E}_i$  the fact that expectations are firm-specific.

After discounting and summing the stream of expected profits over the planning horizon *S* as shown in equation (1) and given a specific vector of expected high-carbon capacity utilisation rates  $\{u_s\}_t^S$ , an individual firm obtains values for the expected return rates of its two technological options,  $\mathbb{E}_i(r_H)$  and  $\mathbb{E}_i(r_L)$ . Defining

$$\mathbb{E}_{j}(\varphi) = \mathbb{E}_{j}(r_{L}) - \mathbb{E}_{j}(r_{H}) \tag{5}$$

as the difference between the two expected return rates, firm *j* invests in a unit of  $K_L$  if  $\mathbb{E}_j(\varphi) > 0$  or in a unit of  $K_H$  if  $\mathbb{E}_j(\varphi) < 0$ .

#### 2.2. Stranding expectations

At the aggregate level, we assume stranding expectations to be heterogeneous across firms, normally distributed around a central expectation path and serially uncorrelated. This choice is motivated by the additive stability of independent normal distributions and the easy interpretation of parameters.<sup>6</sup>

Formally, this writes:

$$\mathbb{E}_{j}(u_{H,s}) = u_{H,s}^* + \varepsilon_{u,j,s},\tag{6}$$

where  $u_{i,s}^*$  identifies a benchmark 'central stranding' expected path and the error term  $\varepsilon_{u,j,s}$  represents the idiosyncratic expectation of firm *j*. Its distribution  $\varepsilon_{u,s}$  represents the diversity of expectations. It follows a normal distribution with mean 0 and variance  $\sigma^2$ , *i.e.*  $\varepsilon_{u,s} \sim \mathcal{N}(0, \sigma_{u,s})$ . We call this schedule 'central stranding' in that it would be the path expected if agents did not have idiosyncratic beliefs. This path can be considered the ongoing 'common wisdom' on the market that serves as a focal point for agents (Schelling, 1960). This 'common wisdom' can take the form of a general narrative that most agents believe to be true and that guides their investment decisions (Boyer, 2018). It can also be construed as the government's policy objective that agents use as an anchor for their beliefs. However, we refrain from adopting this definition, since we do not model the interactions between regulator credibility and investment behaviours, which may have critical implications (see Campiglio et al., 2023, on this topic). For instance, a very ambitious policy plan may be seen as non-credible, and increase belief heterogeneity. As a result, we interpret central expectations only as market norms.

#### 2.3. Central stranding expectation

Three crucial factors contribute to determining firms' transition expectations: i) expected growth of demand; ii) expected speed of development of new technologies; iii) expected long-term share of the new technology in the mix. We analyse each of them in turn.

First, firms expect a constant and positive growth rate  $g_E$  of electricity demand within their planning horizon. This assumption is supported by energy demand forecasts, which argue that the increase in global population and the economic development of emerging economies require an expanding supply of electricity (Enerdata, 2021). An expansion of electricity production is also commonly perceived as a crucial component of decarbonisation strategies.

Second, firms expect the transition to follow an S-shaped curve, as technological transitions typically exhibit this pattern (Fouquet, 2010; Grubb et al., 2020). New technologies often first emerge as niches within a technological paradigm dominated by the incumbent technology (in this case: fossil-based capital stocks). After some early adoption, expansion can accelerate due to a number of factors, including the decline in production costs, the diffusion of information and the growing social legitimisation of the technology (Geels, 2002; Geroski, 2000). In the electricity sector, network effects also play a role in facilitating adoption, as complementary infrastructure and policy develop to better integrate renewable technology. This growth is however limited by factors such as market saturation and physical capacity and slows down as we approach the 'carrying capacity' of the system. Expectations of low-carbon energy share  $\ell_E = \frac{e_L}{e_L + e_H}$  thus move logistically in psychological time s:

<sup>&</sup>lt;sup>5</sup> For the sake of brevity, we conflate in what follows the profit rate and the retained profit rate.

<sup>&</sup>lt;sup>6</sup> We also implement the model with another family of addition-stable distributions, Stable laws, whose skewness and kurtosis can be parameterised and of which the Normal distribution is a special case. The qualitative insights are similar. However, as Appendix A discusses, the properties of non-normal Stable distributions render the interpretation of results more uncertain. More general results could be found numerically using convolution products; however, we prefer to opt for addition-stable distributions for analytical clarity. See also Appendix B, where we extend the model by allowing for beliefs on the future profitability of both technologies.



**Fig. 2.** Illustration of central expectations on future diffusion of low-carbon energy and the associated high-carbon capital utilisation rate. (The logistic function defined in equation (5) has a lower asymptote of zero. In the chart, we only show the portion of the curve to the right of our starting calibration value for the share of low-carbon energy,  $\ell_{E0} = 0.22$ .)

$$\mathbb{E}(\ell_{E,s+1}) = \mathbb{E}(\ell_{E,s}) \left[ 1 + b_{\ell} \left( 1 - \frac{\mathbb{E}(\ell_{E,s})}{\mathbb{E}(\bar{\ell})} \right) \right],\tag{7}$$

where  $b_{\ell}$  represents the unconstrained expected speed of transition, *i.e.* the growth rate of  $\ell_E$  when close to its lower asymptote and  $\bar{\ell}$  represents the maximum expected share of low-carbon technologies.

Third, firms expect a less-than-full energy transition. Mainly due to the current lack of reliable large-scale energy storage technologies, some non-renewable generation capacities — able to adjust their output more rapidly than renewable technologies — will likely still be needed to deal with demand peaks. In addition, firms might also include radical technological breakthroughs in their mental scenarios (*e.g.* carbon capture and storage, direct air capture, geoengineering), which might result in an early deceleration of  $\ell_E$ . Hence,  $\mathbb{E}(\tilde{\ell})$  is lower than 1.

Hence, given a set of expectation parameters  $g_E$ ,  $b_{\ell'}$  and  $\bar{\ell'}$ , the rational stranding path  $\{u_s^*\}_t^S$  identifies the succession of expected capital utilisation rates for which: i) positive high-carbon investments  $I_H$  are expected to the amount necessary to provide the exact amount of  $K_H$  needed to satisfy demand  $e_s^d - e_{L,s}$  at a full capacity utilisation rate  $u_H^f$ , which we call  $K_H^d$ ; ii) no premature decommissioning or technological re-conversion of capital stocks is desired by firms; *i.e.* firms do not expect negative investments. We can thus define central stranding expectations of high-carbon investments  $I_{H,s}^*$  as:

$$I_{H,s}^* = Max \left[ K_{H,s}^d - (1-\delta) K_{H,s}^*; 0 \right].$$
(8)

Given (7) and the definition of  $u_H$  given in section 2.1, we can thus write the resulting rational stranding capacity utilisation  $u_H^*$  as:

$$u_{H,s+1}^{*} = \frac{(1+g_{E})\left(e_{s}^{d}-e_{L,s}\left[1+b_{\ell}\left(1-\frac{e_{L,s}}{\bar{\ell}e_{s}^{d}}\right)\right]\right)}{\xi_{H}[(1-\delta)K_{H,s}+I_{H,s}^{*}]}.$$
(9)

Fig. 2 portrays how the expected share of low-carbon energy  $\ell_E$  and the benchmark expected high-carbon capital utilisation rate  $u_H^*$  move in psychological time *s* for different values of expected intrinsic speed of renewable development  $b_{\ell}$ .

It is worth noting that these expectations concern the whole of the high-carbon sector. We make the assumption that utilisation is homogeneous across high-carbon capital vintages, such that any future decrease in utilisation rates at the sector level will translate one-to-one to individual capital units to be installed today. Hence, if agents expect a lower utilisation rate at the sector level, they will expect a lower utilisation rate for their prospective new capacity.<sup>7</sup>

Stranding expectations are key in determining investment behaviours, as per Proposition 1 below:

**Proposition 1.** For large enough values of  $b_{\ell}$ , there exists an interval  $T \in [|0, S|]$  such that,  $\forall s \in T$ ,  $\pi_{L,s} - \pi_{H,s} > 0$ .

**Proof.** Demonstration given in Appendix F.1.

In other words, if they suppose that the development of low-carbon energy will be fast, agents expect at least a period of psychological time over their planning horizon over which, in the future, low-carbon technologies will be more profitable than high-carbon technology, which could induce them to change their investment behaviour.

<sup>&</sup>lt;sup>7</sup> This simplifying assumption could be challenged on the ground that older units will be under-utilised first. However, as noted by prospective studies (Grant and Coffin, 2020), high-carbon capital installed today will anyways have to suffer significant under-utilisation or premature decommissioning, even if older units stopped in priority.



Fig. 3. Evolution of disagreement in psychological time - Source: AR6 Scenario Explorer and own calculations. Each point is the variance of the share of renewable energy across all scenarios provided by the AR6.

#### 2.4. Heterogeneity of stranding expectations

Heterogeneity in expectations is represented by the error term  $\varepsilon_{u,s} \sim \mathcal{N}(0, \sigma_{u,s}^2)$  of Equation (6), where  $\sigma_{u,s}$  is a parameter indicating the strength of expectation diversity. The lower  $\sigma_{u,s}$ , the more stranding expectations are homogeneous and close to  $u_{H,s}^*$ . The higher  $\sigma_{u,s}$ , the more stranding expectations are diverse and possibly far away from  $u_{H,s}^*$ .

An important question is how this dispersion should change in time s. The literature on projection disagreements amongst economic forecasters provides inconclusive insights on the term structure of opinion diversity. Binder et al. (2022) show that disagreements largely depend on the variable at stake. The term structure for some variables (growth rate, inflation) show higher (lower) disagreement over short-run (long-run) forecasts. For other variables (unemployment, base rates), the disagreement's term structure is strictly increasing. Patton and Timmermann (2010) show a logistically increasing uncertainty disagreement for all the variables they consider. However, these studies focus on macroeconomic, short-run macroeconomic variables. Inter-model comparison exercises using Integrated Assessment Models (IAMs) focus instead on more relevant variables for our purposes, such as energy capacity shares. Kriegler et al. (2014) show that technology deployment schedules can vary importantly across models, which can be taken as a measure of disagreement along the term structure.

To build on this insight, we consider projections of the share of renewables in total electricity production across all IPCC (2022) scenarios<sup>8</sup> and compute their variance at each available simulation step over the 2020-2060 period.<sup>9</sup> We consider IPCC scenarios as a good proxy for expectations, in that they provide long-run projections that policymakers and economic agents can take as a benchmark in forming their own expectations. We assume further that the cross-scenario variance in the share of renewables reasonably proxies disagreement across agents, who can give more or less credence to one or the other IPCC projection. The result of this exercise is displayed in Fig. 3.

As can be seen, the variance follows a near-perfect sigmoid pattern, on which we fit a logistic function with an intrinsic growth rate of 0.28. Hence, we let  $\sigma$  increase in psychological time *s* following a logistic pattern:

$$\sigma_{u,s+1} = \sigma_{u,s} \left[ b_{\sigma} \left( 1 - \frac{\sigma_{u,s}}{\bar{\sigma}_{u}} \right) \right], \tag{10}$$

where  $b_{\sigma}$  is the unconstrained growth rate of  $\sigma_{u,s}$ ,  $\bar{\sigma}_u$  represents the maximum heterogeneity in the long run and  $\sigma_u = \sigma_0$  at time *t*. Short-term views regarding the transition are roughly aligned, as agents observe the current state of things and recent trends. In other words, given the past evolution of low-carbon energy production shares up to time *t*, the expected low-carbon energy share for *t* + 1 will be rather homogeneous across firms. However, transition expectations then diverge rapidly over the medium term, capturing the heterogeneity of opinions concerning technological and policy prospects. While some might be expecting decarbonisation to unravel rapidly in the course of the next decade, others might expect fossil fuels to remain the backbone of the global economy in the decades to come. In the longer run, the marginal divergence of expectations weakens, approaching a fixed maximum level  $\bar{\sigma}_u$ . In other words, the diversity of long-term opinions remains roughly constant once a certain beyond a certain threshold in psychological time.

<sup>&</sup>lt;sup>8</sup> IPCC scenarios are available at https://data.ece.iiasa.ac.at/ar6.

<sup>&</sup>lt;sup>9</sup> The picture over 2020-2100 is similar, with the variance of projections oscillating around a carrying capacity.



Fig. 4. Stylised representation of our approach to belief heterogeneity. To each time  $s \in [|1, S|]$  corresponds a central utilisation expectation. Each bell curve represents the actual distribution of expectations (rotated 90°), with time-varying variances and means. Here displayed an example for  $\ell = 0.9$ ,  $b_{\ell} = 0.2$ ,  $\sigma_0 = 0.01$  and  $\bar{\sigma} = 0.5$ .

#### 2.5. Aggregate investment decisions

We now want to calculate the share of aggregate investments flowing into each technology. We define  $\ell_I = \frac{I_L}{I_H + I_L}$  as the share of total investments allocated to low-carbon capital stocks. In our setting, this is equivalent to the probability for an individual firm to obtain a positive  $\varphi$ ; that is, to expect the low-carbon return rate  $r_L$  to be higher than the high-carbon return rate  $r_H$ .

Given our assumption of a normal distribution of the error term  $\epsilon_u$  in equation (6) and considering that linear transformations maintain the normal distribution pattern, we can rewrite  $\varphi$  as the sum of a deterministic 'rational stranding' component  $\varphi^*$  and an error term  $\varepsilon_{\omega} \sim \mathcal{N}(0, 1)$ :

$$\varphi^* = \frac{R^*}{\Gamma},\tag{11}$$

with:

$$R^* = \sum_{s=0}^{S} \beta^s \left\{ p_E \xi_L - \alpha_L \psi_L c_L - \left[ \left( p_E - \frac{p_F}{\xi_F} \right) \xi_H u_{H,s}^* - \alpha_H \psi_H c_H \right] \right\}$$
(12)

$$\Gamma = \sqrt{\sum_{s=0}^{S} \beta^{2s} (p_E - \frac{p_F}{\xi_F})^2 \sigma_{u,s}^2}.$$
(13)

All details of the transformation are discussed in Appendix D. Fig. 4 schematises our representation of belief heterogeneity as a set of normal distributions with varying means and variances.

 $R^*$  can be thought of as a 'benchmark return rate', which would prevail in the absence of idiosyncratic beliefs  $\varepsilon$ , *i.e.*, when agents' expectations are aligned on the central projection. In the presence of belief dispersion, it represents the average return expectation.  $\Gamma$ is a measure of the extent of belief dispersion across the planning horizon. In the definition of  $\phi^*$ ,  $\Gamma$  can be taken as a belief-correction term used on  $R^*$  to account for the extent of belief dispersion. A high  $\Gamma$  implies that agents form beliefs that can be significantly different from the average expectations, making  $R^*$  less important for aggregate behaviour.

We can then derive  $\ell_I$ , our main variable of interest, as the value of the cumulative distribution function  $\Phi(\cdot)$  of the standard normal distribution at  $\varphi^*$ . That is,

$$\ell_I = \Pr(\epsilon_{\varphi} < \varphi^*) = \Pr(r_L > r_H) = \Phi(\varphi^*) \tag{14}$$

To allow ourselves to explore large  $\sigma_{u,s}$  values while keeping heterogeneous beliefs realistic, we censor the  $\varepsilon_{u,s}$  distributions above 1 and below zero. After some transformations (see Appendix D and E), the final definition of  $\ell_I$  is:

$$\ell_I = \frac{\delta_{\varphi_0 \varphi^*}(\Phi(\varphi^*) - \Phi(\varphi_0^*)) - \delta_{\varphi_1 \varphi^*}(\Phi(\varphi_1^*) - \Phi(\varphi^*))}{\Phi(\varphi_1^*) - \Phi(\varphi_1)},\tag{15}$$

where  $\delta_{uv} = \begin{cases} 1 & u < v \\ 0 & u \ge v \end{cases}$ and  $\varphi_0^*$  and  $\varphi_1^*$  are the values  $\varphi$  would take if  $\mu_{H,s}^* = 0 \,\forall s \text{ or } \mu_{H,s}^* = 1 \,\forall s \text{ respectively. For the values of } \varphi^*$  we are exploring, we are always in a case in which  $\varphi_0 \le \varphi^* \le \varphi_1$ , such that the equation reduces to:

$$\mathcal{C}_I = \frac{\Phi(\varphi^*) - \Phi(\varphi_0)}{\Phi(\varphi_1) - \Phi(\varphi_0)} \tag{16}$$

Symbol	Variable	Value	Source
Production			
e <sub>d</sub>	Initial energy production	3243 GWh	European Commission (2019)
$K_{H,0}; K_{L,0}$	Initial capital stock	659 GW; 288 GW	European Commission (2019)
$\xi_H; \xi_L$	Capital productivity	4.5158; 2.9145	Eurostat (2022)
$u_{H}^{f}; u_{I}^{f}$	Full capacity utilisation rate	0.85	Eurostat (2022)
$u_H, u_L$	Initial utilisation rate	0.85	Eurostat (2022)
$p_{e}$	Price of electricity	0.2 bn\$/TWh	Eurostat (2022)
ξŗ	Productivity of fossil fuels	0.114	EIA (2020)
$p_f$	Price of fossil fuels	0.0022 bn\$/TWh	Eurostat (2022)
$\delta_H; \delta_L$	Capital depreciation rate	0.03; 0.04	IEA (2020d)
$c_{H}^{K}; c_{I}^{K}$	Capital cost	3.5 bn\$/GW; 2.9 bn\$/GW	IEA (2020d)
g <sub>e</sub>	Energy demand growth	0.012	Enerdata (2021)
Finance			
$\psi_H; \psi_L$	Debt-to-investment ratio	0.7; 0.75	Baruya (2017); IRENA (2020)
$\iota_H; \iota_L$	Interest rate on loans	0.045; 0.0394	Kempa et al. (2021)
$LT_H; LT_L$	Loan tenor	15 years	Refinitiv (2022)
Belief heterogeneity			
$b_{\sigma}$	Intrinsic growth rate for $\sigma_{\mu s}$	0.28	IPCC (2022)

Table 1 Technological and financial parameter

#### 3. Calibration

This section presents our calibration strategy. The model's time step in *s*-time is explicitly yearly. All monetary values are expressed in billion 2019\$US; electricity production in terawatt-hours (TWh); electricity generation in gigawatts (GW); fossil energy in British thermal unit (Btu). We use the EU27 region in 2019 as our model economy. We can distinguish two categories of parameters. First, a set of technological or financial parameters calibrated to replicate empirical evidence and summarised in Table 1. Second, a set of behavioural parameters for which, in absence of solid empirical foundations, we perform a sensitivity analysis along reasonable value ranges in Section 4.

# 3.1. Technological and financial parameters

We include solar, wind and other minor renewable energy technologies in our low-carbon category and all other technologies in the high-carbon category. While hydropower and nuclear can be considered to be low-carbon considering their low emission intensity, additional environment-related issues (*e.g.* land use, waste treatment) usually lead them to be treated separately from wind, solar, geothermal and other renewable technologies. In addition, we take into consideration the fact that installing large hydro plants is now mostly infeasible in Europe, where generation of electricity from hydro has been stationary since 2000, nor strongly desired by policy-makers due its environmental footprint. Furthermore, across scenarios, uncertainty revolves mostly around the share of 'modern' renewables compared to other energy sources (Tsiropoulos et al., 2021). We thus focus on them and include nuclear and hydro in our 'high-carbon' category. We investigate alternative categorisations, with hydro and nuclear included as part of the low-carbon technology, in Appendix C.

We set the initial values of capital stocks following the European Commission (2019), which reports a total installed capacity of approximately 947 GW in 2019 for the EU-27 region, of which around 30% is made of solar, wind or geothermal plants. We thus set  $K_{L,0} = 288$  GW and  $K_{H,0} = 659$  GW, which implies that the share of low-carbon capital  $\ell_{K,0} = \frac{288}{947} \approx 30\%$ .

We further make a difference between utilisation rates and capacity factors. Utilisation rates are demand-driven and depend on economic factors. They refer to the degree to which the capital stock is used and are measured as the ratio between production and potential production at the best of technical possibilities. It is common for firms to maintain their available capital stocks operating on average at a rate lower than 100%, to be able to accommodate for peaks in demand. The normal capacity utilisation rate  $u^f$  is equal to 0.85 for both technologies, which is roughly in line with utilisation rates at the macro level in the European Union in 2019 given by Eurostat. We assume that the economy starts from a situation where both technologies operate at normal capacity and that firms expect the current-period utilisation to be equal to normal *i.e.*, we set  $u_{L,0} = u_{H,0} = 0.85$ .

On the other hand, capacity factors represent the technical limitations preventing an electricity-generation technology to operate at 100%. We assume them to be encapsulated in our productivity factors  $\xi_H$  and  $\xi_L$ , in the sense that they are purely defined by technology. Given our assumptions on utilisation rates, we calibrate our productivity parameters  $\xi_H$  and  $\xi_L$  to match energy production from high- and low-carbon technology in Europe. The share of wind, solar and geothermal technologies in total gross electricity production in Europe was around 22% in 2019. With a utilisation rate of 85% and a total energy demand equal to 3243 TWh this yields  $\xi_L = \frac{0.22*3243}{0.85*628} = 2.9145$  and  $\xi_L = \frac{0.78*3243}{0.85*629} = 4.5158$ . Given that a GW of capacity would produce 8.76 TWh in a year at full capacity, this yields implicit capacity factors of  $\frac{2.9145}{8.76} \approx 33\%$  and  $\frac{4.5158}{8.76} \approx 51\%$ , which are both roughly in line with 2018 capacity factors (*i.e.* uncorrected for utilisation rates in Europe) (IEA, 2019).

Та	ble	2
----	-----	---

Behavioural parameters.				
Variable	Meaning	Reference value(s)	Sensitivity range	
S	Length of planning horizon	20	[2, 40]	
$b_{\ell}$	Expected intrinsic growth rate for $\ell$	0.15; 0.25	[0, 0.5]	
Ī	Maximum expected $\ell$	0.85	[0.5, 0.95]	
$\sigma_0$	Opinion diversity at time t	0.01	[0, 0.1]	
$\bar{\sigma}$	Maximum opinion diversity	0.5	[0.01, 2]	
ρ	Corporate discount rate	0.05	[0.01, 0.1]	

The data on electricity prices for both households and non-household consumers is provided by Eurostat,<sup>10</sup> and shows values ranging from 0.005 to  $0.27 \notin$ /KWh in 2019 for the EU27 region. We adopt a middle value by setting  $p_F = 0.2$  bn\$/TWh.

Fossil productivity parameter  $\xi_F$  transforms fossil fuels, expressed in trillion British thermal units (Btu), into electricity, expressed in TWh. 1 TWh physically corresponds to approximately 3.5 trillion British thermal units (Btu). We then need to adjust this number for the efficiency of thermal plants (heat rate), to account for energy losses arising from the conversion process. According to EIA (2020), the efficiency of thermal plants is around 0.33-0.45. Taking an intermediate value of 0.4, we calculate EU thermal plants to require 8.75 (= 3.5/0.4) trillion Btu to produce 1 TWh of electricity. The  $\xi_F$  coefficient is then computed as the inverse of this number, that is  $\frac{1}{8.75} \approx 0.114$ .

The price of natural gas in 2019 for non-household consumers in the EU27 region was around  $0.03 \in /KWh$ .<sup>11</sup> Rescaling to Btu and taking into consideration that the price of coal is usually lower than the one of gas, we set  $p_F = 0.0025$  bn\$/tnBtu.

We set depreciation rates  $\delta_H$  and  $\delta_L$  as the inverse of technology-specific asset lifetimes. IEA (2020a) reports expected lifetimes of 25 years for solar and wind plants and 30-40 years for fossil-fuelled plants. We thus set  $\delta_H = 0.03 \approx 100/33$  and  $\delta_L = 0.04 = 100/25$ .

Capacity installation cost parameters  $c_i$  represent the cost of installing a unit of generating capacity (the 'overnight constructions costs'). IEA (2020a) describes the overnight cost for various technologies hypotheses made by the IEA in its projection exercises.<sup>12</sup> By taking a weighted average based on the European energy mix, we retain  $c_H^k = 3.5$  bn\$/GW and  $c_L^k = 2.9$  bn\$/GW.

Financing costs depend on the loan interest rate  $\iota_i$ . Recent findings from Kempa et al. (2021) show an average 3.5% spread to Libor for high-carbon projects and that renewable projects face a 16% lesser markup. We then assume, for Europe, a 1% risk-free interest rate, resulting in  $\iota_H = 0.01 + 0.035 = 0.045$  and  $\iota_L = 0.01 + 0.035 \times 0.84 \approx 0.0394$  Debt-to-investment ratios ( $\psi_H$ ,  $\psi_H$ ) for renewable projects are taken from IRENA (2020), which reports a 75% rate. For high-carbon energy sources, sources are more conflicted (Baruya, 2017), with numbers ranging between 60 and 80% debt-financing. We thus adopt a middle-range assumption of a 70% rate. Regarding the loan term of debt financing, we rely on the Refinitiv project database. Albeit scarce, data shows a rough average of a 15-year loan term for all technologies (LT = 15).

Finally, we take energy demand growth from EnerData Enerblue scenario (Enerdata, 2021), which projects a 43% increase in final electricity consumption demand between 2020 and 2050. Assuming a constant growth rate over this period yields a yearly 1.2% growth.<sup>13</sup>

## 3.2. Behavioural parameters

Our model includes several parameters capturing the expectations and behavioural features of investment decision-makers. Despite the recent emergence of a stream of research contributions trying to assess climate-related expectations via surveys or financial econometrics (Bolton and Kacperczyk, 2020; Krueger et al., 2020), we currently do not have reliable data on which to calibrate these parameters. An exception is  $b_{\sigma}$ , which we calibrate to 0.28 based on our analysis of IPCC scenarios (see Section 2.4). However, since uncertainty acts on a different compact, we keep  $\sigma_0$  and  $\bar{\sigma}$  free.

For this reason, we illustrate our analytical results with sensitivity analyses on pairs of parameters while keeping the rest of the behavioural parameters fixed. We thus choose a reference value (to be kept constant while the parameter is not part of the sensitivity analysis) and a sensitivity range for each parameter. Since we are interested in exploring how these behavioural parameters could affect investment choices, we keep the sensitivity ranges large enough to capture all possible dynamics. In particular, we choose the sensitivity range for the maximum expected share of low-carbon technologies  $\ell^{\vec{e}}$  and the intrinsic growth rate of the expected low-carbon energy share  $b_{\sigma}$  to match scenarios for the European Green New Deal (Tsiropoulos et al., 2021). Table 2 offers a summary of our choices.

#### 3.3. High- and low-stranding projections

We conclude our calibration by specifying two representative central projection paths. In both of them, the long-run share of low-carbon energy is left fixed at 90%, in the upper distribution of the European Commission's scenarios (Tsiropoulos et al., 2021),

 $<sup>^{10} \ \ \</sup>text{See series nrg_pc} \ \ \text{204 and nrg_pc} \ \ \text{205 available at } \ \ \text{https://ec.europa.eu/eurostat/data/database.}$ 

<sup>&</sup>lt;sup>11</sup> See Eurostat series nrg\_pc\_203\_c.

<sup>&</sup>lt;sup>12</sup> For hydro, the IEA does not provide data specific to Europe; we thus relied on a weighted average of costs across countries (IEA, 2020b).

<sup>&</sup>lt;sup>13</sup> We follow most of the energy modelling literature in assuming that, while electricity demand can be affected by energy efficiency and other demand-side measures, the composition of its supply does not significantly affect the amount of power demanded by economic agents.

(18)

Table 3					
Intrinsic growth rate values for low and					
high-stranding central expectations.					

Scenario type	$b_\ell$
Low stranding	0.15
High ambition	0.25

so to leave the intrinsic growth rate as the only degree of freedom. In formal terms, low-stranding projections can be defined as the set of  $b_{\ell}$  such that  $R^* < 0$  for our benchmark *S* and  $\rho$ . Reciprocally, high-stranding projections can be seen as the set of  $b_{\ell}$  such that the benchmark return rate  $R^*$  is strictly positive. We choose  $b_{\ell} = 0.15$  for a low-stranding central projection and  $b_{\ell} = 0.25$  for a high-stranding central projection. We summarise this choice in Table 3. Whenever relevant, we will compare results from these two central projections.

We characterise a low-stranding scenario as a gradual transition, with a slow development of low-carbon technologies that does not result in much stranding. High-carbon energy sources remain operational for relatively long and be decommissioned gradually. By contrast, a high-stranding central expectation would feature a quicker deployment of low-carbon technology, resulting in larger stranding. This narrative implies a more disruptive transition, in which high-carbon technology is quickly replaced in the short run (Grubb et al., 2020). Our 'low-stranding' scenario can be roughly compared to the IEA's 'Announced Pledges' scenario (IEA, 2022a), in which fossil fuels are slowly replaced by alternative technologies. The IEA's 'Net Zero by 2050' scenario (IEA, 2022b) entailing a more rapid and intense low-carbon transition, could be compared instead to our 'high-stranding' path.

#### 4. Results

This section expounds our main takeaways. We start by studying how belief heterogeneity affects aggregate investment decisions. To do so, we begin in Section 4.1 with a polar case without belief heterogeneity, corresponding to a benchmark in which agents' expectations are fully coordinated around the central expectation. We then lift the no-heterogeneity assumption to explore how our modelling proposal influences results compared to the benchmark. Subsequently, in Section 4.2, we explore how aggregate investment behaviours change with various levels of belief heterogeneity. We notably characterise the relative effect of short ( $\sigma_0$ ) and long-run ( $\bar{\sigma}$ ) belief heterogeneity and derive some properties in the case of hyperbolic heterogeneity. Sections 4.3 and 4.4 explore instead the interactions between belief heterogeneity and preference for the present: the discount rate  $\rho$  and the length of the planning horizon *S*.

#### 4.1. Introducing belief heterogeneity

We start by assuming  $\sigma_{u,s} = 0 \forall s \in [|1, S|]$ , *i.e.*, we describe a situation in which all agents believe in the central projection. Being this a limit condition on our model, we can prove Proposition 2:

**Proposition 2.** For  $\sigma_{u,s} = 0 \forall s \in [1, S]$ ,  $\ell_I$  tends towards a degenerate probability distribution function, whereby:

	0	$if R^* < 0$	
$\ell_I = \langle$	0.5	$if R^* = 0 \tag{(1)}$	[17)
	1	$if R^* > 0$	

**Proof.** Demonstration is given in Appendix F.2.

In other words, without belief heterogeneity the system can only achieve three outcomes: i) the whole populating invests in low-carbon energy ( $\ell_I = 1$ ); ii) none of it does ( $\ell_I = 0$ ); or iii) exactly half of the population does ( $\ell_I = 0.5$ ). The interpretation is straightforward: absent belief heterogeneity, only the central projection, which rules the sign of  $R^*$ , matters. If this implies that low-carbon sources will be more profitable over the planning horizon according to this central projection ( $R^* > 0$ ), then it is rational for all investors to invest in low-carbon energy. The intermediate case  $R^* = 0$  would denote a 'total indecision' situation, in which agents are indifferent between the two technologies and therefore exhibit a 50-50 dispatch in aggregate. We illustrate these findings in Fig. 5, Panel (a).<sup>14</sup>

The value of  $R^*$  depends on the discount rate  $\rho$ , the planning horizon *S* and on the variables linked to the central expectations, *i.e.* the maximum expected share of low-carbon technologies  $\ell^{\bar{\ell}}$  and the intrinsic growth rate  $b_{\ell}$ . More precisely, we rewrite:

$$R^* = R^*(\rho, S, (u_{H,s})_{s \in [|1,S|]})$$

 $^{14}~$  Note that the case  $\ell_I=0.5$  does not show because of the numerical simulation steps used for the chart.

Journal of Economic Behavior and Organization 216 (2023) 535-567



Fig. 5. Effect of belief heterogeneity.

Based on the definition of Equation (12), it can be shown  $\frac{\partial R^*}{\partial \rho} < 0$ . This is a standard effect of increasing the discount rate. It leads agents to weigh less later periods over which stranding is expected to be strong enough to give an edge to low-carbon sources. For a high  $\rho$ ,  $R^*$  can be negative.

Further, the lower the  $(u_{H,s})_{s \in [1,S]}$  schedule, the lower  $R^*$ . The position of the  $(u_{H,s})_{s \in [1,S]}$  schedule depends directly on the  $(\ell_{E,s})_{s \in [1,S]}$  schedule, governed by  $\bar{\ell}$  and  $b_{\ell}$ , as per Proposition 3 below:

**Proposition 3.** Consider a logistic sequence  $x_n = x_{n-1} \left(1 + b(1 - \frac{x_{n-1}}{K})\right)$  where K is a carrying capacity and b an intrinsic growth rate. Consider then  $x_0$  the first term of this sequence. It follows that,  $\forall n$  such that  $x_n < K$  and  $b \le 1$ :

$$\frac{\partial x_n}{\partial K} \ge 0 \tag{19}$$

$$\frac{\partial x_n}{\partial b} \ge 0$$
(20)
$$\frac{\partial x_n}{\partial x_0} \ge 0$$
(21)

With the last proposition holding for  $x_0 < K$ .

۵.,

# **Proof.** Demonstration is given in Appendix F.3.

For what concerns the expected low-carbon share of energy, this implies that  $\forall s$ :

$$\frac{\partial \ell_{E,s}}{\partial \bar{\ell}} \ge 0 \tag{22}$$

$$\frac{\partial \ell_{E,s}}{\partial b_s} \ge 0 \tag{23}$$

The intrinsic growth rate  $b_{\ell}$  rules the number of periods in time *s* needed to reach the maximum share  $\ell$  in expectations. If  $\ell$  is increased with  $b_{\ell}$  held constant, agents will expect that a higher maximum share will be reached within the same amount of time. It requires  $\ell_s$  to be higher or equal  $\forall s$ . Conversely, given a certain  $\ell$ , a higher  $b_{\ell}$  means that less time is required to reach  $\ell$ . This entails a steeper growth and higher  $\ell_s \forall s$ . Hence, a path with higher  $\ell$  or  $b_{\ell}$  entails more stranding expectations due to a speedier increase of the share of low-carbon energy, which will result in higher low-carbon investment.

In economic terms, this proposition demonstrates that, under reasonable assumptions, if agents expect a higher long-term share of renewables  $\ell$  or a speedier transition  $b_{\ell}$ , they will expect a higher share of renewables for all periods within their planning horizon. Given that stranding expectations are directly governed by the expected share of low-carbon investment, this implies that a higher  $\ell$  or  $b_{\ell}$  will imply higher stranding for all periods within the agents' planning horizon.

Finally, as per the definition of  $R^*$ , a minimum planning horizon is required for low-carbon investment to emerge. Indeed, if agents are so short-sighted to only account for early periods, over which stranding is low, they will expect a negative  $R^*$ . However, very long planning horizons are not necessarily improving. Given that stranding is transitory in our model — as high-carbon capital naturally depreciates — the period over which low-carbon capital becomes more profitable is finite. Once it is over, agents will expect an edge for high-carbon technologies again over subsequent periods, which may compensate for the positive payoff over the stranding period. Hence the following Proposition:

**Proposition 4.** It is possible to define an interval  $S = [|\underline{S}; \overline{S}|] \subset [|1, S|]$  such that, for a given  $\rho$ ,  $b_{\ell}$  and  $\overline{\ell}$ ,  $R^* > 0 \forall s \in S$  and  $R^* \leq 0$  otherwise. S can be empty.

**Proof.** Demonstration is given in Appendix F.4.



**Fig. 6.** Long-run and short-run belief heterogeneity. Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.85" on Panel (a) corresponds to all  $(\bar{\sigma}, \sigma_0)$  combinations for which the share of low-carbon investment  $\ell_I$  is equal to 0.85 in the case of high stranding central expectation.

This proposition states that, over their planning horizon, agents will expect low-carbon energy sources to be more profitable than high-carbon technologies over a sub-period *S*, typically the period over which the stock of high-carbon capital will suffer most from stranding.

We now introduce belief heterogeneity, by assuming the shape for the  $\sigma_{u,s}$  schedule presented in Equation (10). Results are illustrated in Fig. 5, Panel (b).

As can be seen, the bang-bang solution disappears to give rise to a whole gradient of interior solutions ranging between 0 and 1 depending on the value of parameters  $b_{\ell}$  and  $\bar{\ell}$ . The smoothness of the corresponding surface depends positively on belief heterogeneity.

Lifting the limit condition implies that the variables linked to relative return rates,  $\varphi^*$ ,  $\varphi_0$  and  $\varphi_1$ , take finite value and that the  $\ell_I$  function takes value over the whole [0, 1] interval. Intuitively, introducing belief heterogeneity allows for the existence of investors whose behaviour will drift from the central projection. Hence, for a projection entailing zero investment in the no-heterogeneity case, introducing idiosyncratic beliefs is equivalent to assuming that a part of the population will expect sufficient stranding to invest in low-carbon energy.<sup>15</sup> Reciprocally, for a high-stranding central projection, entailing 100% low-carbon investment in the absence of heterogeneity, allowing for a population of norm-contrarians will entail less low-carbon investment than in a no-heterogeneity case, since this share of the population will expect insufficient stranding.

Including a degree of belief heterogeneity increases significantly the indeterminacy of the model. This result is of important policy relevance, as the results implied by different degrees of belief heterogeneity can be very different from those flowing from an assumption of perfectly aligned expectations. We explore the impact of various levels of belief heterogeneity in the following and their policy consequences.

#### 4.2. Belief heterogeneity, investment decisions and the central projection

To explore the role of belief heterogeneity, we let our parameters  $\sigma_0$  (short-run belief dispersion) and  $\bar{\sigma}$  (long-run belief dispersion) vary. Proposition 3 implies that:

$$\frac{\partial \sigma_{u,s}}{\partial \sigma_0} \ge 0$$
(24)
$$\frac{\partial \sigma_{u,s}}{\partial \bar{\sigma}} \ge 0$$
(25)

As a result, any increase in these two parameters will result in a higher  $(\sigma_{u,s})_{s \in [[1,S]]}$  schedule, which will increase the beliefcorrection factor  $\Gamma$  defined in Equation (12). It is possible to demonstrate the following proposition:

**Proposition 5.** The effect of a higher  $\sigma_0$  or  $\bar{\sigma}$  will depend on the sign of  $R^*$ . If  $R^* < 0$ , then  $\frac{\partial \mathcal{E}_I}{\partial \sigma_0} \ge 0$ ,  $\frac{\partial \mathcal{E}_I}{\partial \bar{\sigma}} \ge 0$ . Furthermore,  $\mathcal{E}_I$  is concave in  $\sigma_0$  and  $\bar{\sigma}$ . If  $R^* > 0$ , then  $\frac{\partial \mathcal{E}_I}{\partial \sigma_0} \le 0$  and  $\frac{\partial \mathcal{E}_I}{\partial \bar{\sigma}} \le 0$ . Plus,  $\mathcal{E}_I$  is convex in  $\sigma_0$  and  $\bar{\sigma}$ . Plus, there exists an  $R' \in [R_0; R_1]$  such that  $\frac{\partial \mathcal{E}_I}{\partial \sigma_0}(R')$  and  $\frac{\partial \mathcal{E}_I}{\partial \bar{\sigma}}(R')$  equal to zero.

**Proof.** Demonstration is given in Appendix F.5.

We further illustrate the effect of belief heterogeneity on  $\ell_I$  in Fig. 6. As a contour plot, it shows isovalue lines, *i.e.*, all the combinations within the  $(\bar{\sigma}, \sigma_0)$  space that yield the same  $\ell_I$  value. The space between the lines gives indication on how the value

<sup>&</sup>lt;sup>15</sup> Due to the symmetry of belief heterogeneity, belief heterogeneity also entails that a population expects *less* stranding than with expectations aligned on the central projection. However, expecting even less stranding than on the central projection does not change the investment choice of these agents. As a result, only the population expecting more stranding has an impact on aggregate investment decisions.



Fig. 7. Evolution of the sensitivity of the share of low-carbon investment  $\ell_I$  to the belief correction factor  $\Gamma$  as a function of the central expectation return spread  $R^*$ .

of the outcome evolves when one of the two parameters is fixed. Typically, an increasing space across isovalue lines for increasing values of a parameter — like in Fig. 6 — would denote a decreasing marginal effect of the moving parameter. In the high-stranding scenario, increasing belief heterogeneity, either in the short or the long run, unambiguously decreases the low-carbon investment share. By contrast, increasing dispersion in the low-stranding case tends to increase low-carbon investment. In both cases, the effect of higher heterogeneity is bounded, as the low-carbon share tends towards a minimum (high stranding) or a maximum (low stranding) as dispersion increases. These results emerge because we define belief heterogeneity as deviations from the central projection or 'market norm'. Hence, the effects of belief heterogeneity on investment decisions also depend on the existing central projection. In the case of low-stranding expectations, 'transition believers' investing in low-carbon projects will be the 'norm contrarians'. Reciprocally, contrarians in the high-stranding case will be 'transition sceptics', who will invest more in high-carbon energy. As a result, belief heterogeneity will have a balancing effect on investment behaviour, the direction of which will depend on the existing norm.

We can also show that the (absolute) effect of more belief heterogeneity on investment allocation is higher when belief heterogeneity is lower. It shows on Fig. 6 through the increasing distance across isovalue lines. This is due to the concavity (in the low-stranding case) or convexity (in the high-stranding case) property of  $\ell_I$  in  $\Gamma$ . Increasing belief heterogeneity from a low-dispersion situation will open a range of stranding expectations close to the cut-off point from which agents switch behaviours. Hence, many switches will occur. Conversely, if belief heterogeneity is high, most beliefs relevant for behaviour switch will already have a sizeable mass in the distribution. Hence a decreasing marginal (absolute) effect of belief heterogeneity.

The previous result, however, only holds for a fixed central projection. This begs studying how changes in belief heterogeneity will affect investment behaviours for various central expectation scenarios.

We first consider the expression of  $\frac{\partial \ell_I}{\partial \Gamma}$  from the demonstration of Proposition 5 (see Appendix F.5, Equation (53)). Then, we fix the belief correction factor  $\Gamma$  (see (12)) to match our benchmark values ( $\sigma_0 = 0.01$  and  $\bar{\sigma} = 0.5$ ). Reducing  $\frac{\partial \ell_I}{\partial \Gamma}$  to a function of  $R^*$ , the function takes the shape displayed in Fig. 7.

Fig. 7 shows that the higher (lower) the curve, the more positive (negative) a change in investment behaviour following a shift in  $\Gamma$  and *vice-versa*. A value of zero would denote that, for the corresponding central projection, moving belief heterogeneity has no effect on investment behaviour, which would be a case of 'perfect resilience'. However, norm resilience is non-monotonic. Sensitivity reaches two optima on each side of the y-axis. As can be seen in Fig. 7, sensitivity is zero for  $R^* = 0$ , *i.e.*, a perfect balance between profit rates at the benchmark. Moving from this point, (absolute) sensitivity increases to a maximum and decreases to perfect resilience.

Policy-wise, this implies that medium-stranding projections are fragile to belief heterogeneity shocks (*e.g.* shocks to confidence; decrease in policy credibility). As a result, policymakers should strive to measure the state of the central expectation to anticipate its resilience to possible shocks in belief dispersion. To reach a given penetration of renewable energy, the regulator may seek to anchor expectations towards higher stranding. It can do so by influencing the central expectation with policy announcements, while making sure to remain credible.

Regarding low-stranding projections, if the market norm is close to business-as-usual (very low stranding), the share of the population expecting enough stranding is so little that marginally increasing it will only affect aggregate investment decisions negligibly. This kind of 'low equilibrium' cannot be changed by a transition-believer minority but only by changing the ongoing market norm.

Another feature emerging from Fig. 6 is that the relative effects of short- ( $\sigma_0$ ) and long-run heterogeneity ( $\bar{\sigma}$ ) depend on the magnitude of one another:

**Proposition 6.** For a given S and  $\rho$ , there exists a threshold ratio  $\left(\frac{\bar{\sigma}}{\sigma_0}\right)^* = \sqrt{\frac{1-exp(-b_{\sigma}S)}{exp(-b_{\sigma}S)}}$  such that  $\left|\frac{\partial \mathcal{E}_I}{\partial \bar{\sigma}_0}\right| > \left|\frac{\partial \bar{\sigma}_I}{\partial \bar{\sigma}}\right| \\\forall \frac{\bar{\sigma}}{\sigma_0} > \left(\frac{\bar{\sigma}}{\sigma_0}\right)^*$ . In other words, the effect of increasing short-run belief dispersion  $\sigma_0$  on investment behaviours is higher than that of increasing long-run dispersion  $\bar{\sigma}$  if  $\bar{\sigma}$  is sufficiently high relatively to  $\sigma_0$ .

**Proof.** Demonstration is given in Appendix F.6.

Whether the condition holds or not depends directly on the length of the planning horizon. Unsurprisingly, the more far-sighted agents are, the more the aggregate investment behaviour will be influenced by belief dispersion on long-run outcomes, unless belief dispersion on long-run developments is already very high.

It is possible to show that this condition holds true for a large proportion of the parameter space we explore. In particular, it holds for our benchmark calibration with S = 20 and  $\sigma_0 = 0.01$  for  $\bar{\sigma} > 0.16$ . It accounts for the whole parameter space we explore in Fig. 6. We therefore focus on this case.

Let us first notice that long-run belief heterogeneity  $\bar{\sigma}$  does not need to be very high for the condition to be fulfilled in our benchmark calibration. Because investment shares are more sensitive to belief heterogeneity for low values if  $\sigma_0$  and  $\bar{\sigma}$ , it entails that, for reasonable values of  $\bar{\sigma}$ , an increase in  $\sigma_0$  can have sizeable effects on aggregate investment behaviours. For high  $\bar{\sigma}$  values, however an increase in  $\sigma_0$ , although it will bear larger effects than an increase in  $\bar{\sigma}$ , will have quantitatively small impacts on  $\ell_I$ . It is due the concavity/convexity properties of  $\ell_I$  in  $\Gamma$  (see above).

The policy implications are nonetheless clear. In the case of a high-stranding central expectation, the priority for policymakers should be to anchor expectations firmly for the earliest periods of the transition, if there is no clear consensus about long-run outcomes. If short-run belief heterogeneity is very low and long-run dispersion even moderately high, slightly higher short-run belief dispersion can have sizeable effects. Anchoring short-term expectations is all the more crucial considering that long-run belief dispersion is to some extent inevitable due to the many uncertainties surrounding the long-run future. Comparatively, aligning expectations on short-term outcomes seems more feasible. In the case of low-stranding expectations, policymakers should be cautious in interpreting positively a large tilt of investment behaviours towards low-carbon investment, as it may represent only a shock to short-run belief heterogeneity. These insights confirm the need for policymakers to carefully measure the state of expectations, in mean and in dispersion.

Fig. 6 further shows that the effect of higher belief heterogeneity dies down as  $\sigma_0$  or  $\bar{\sigma}$  increases, until seemingly reaching a finite value. What happens when the belief-dispersion term  $\Gamma$  tends towards infinity, describing a state of 'full dispersion' of beliefs?

**Proposition 7.** As belief heterogeneity approaches infinity,  $\ell_I$  will tend towards a finite value  $\tilde{\ell}_I = \frac{R^* - R_0}{R_1 - R_0}$ , where  $R^*$ ,  $R_0$  and  $R_1$  are the numerators of  $\varphi^*$ ,  $\varphi_0$  and  $\varphi_1$  respectively. It defines a uniform distribution on  $R^*$  over  $[R_0, R_1]$ . We call  $\tilde{\ell}_I$  'full dispersion' equilibrium and it is a function of  $b_{\ell}$ ,  $\bar{b}$  for a given S and  $\rho$ . Note that if we do not censor our distribution, the 'full-dispersion' equilibrium  $\tilde{\ell}_I$  is always equal to 0.5.

**Proof.** Demonstration is given in Appendix F.7. □

This result means that, if we ensure that expected utilisation rates remain between 0 and 1 (*i.e.*, agents have 'reasonable' beliefs), only the position of the central expectation within the realm of acceptable beliefs matters for aggregate investment behaviour when belief dispersion is very high. Because the distribution of beliefs converges towards a uniform distribution, so does the distribution of expected return rates. As a result, all expectations have the same weight in determining investment decisions as long as they remain within the boundaries of acceptable beliefs. It implies that the only determinant of investment shares will be the position of the central expectation relative to the boundaries defining acceptable beliefs.

This feature has several policy implications. First, it implies that, with high belief heterogeneity, 100% low-carbon investment can only be achieved if the central expectation entails maximum stranding ( $u_{H,s} = 0 \forall s \in [|1, S|]$ ), *i.e.*  $R^* = R_1$ . However, stranding the entire stock of high-carbon capital overnight is a hardly credible scenario. Hence, in the case of a full-dispersion equilibrium, low-carbon investment will always represent less than 100% of the total.

Second, low- and high-stranding projections are asymmetric. High belief heterogeneity for low-stranding projections will always yield *less* low-carbon investment than high belief heterogeneity with high-stranding expectations. This implies that a strong minority of 'high-stranding believers' will never be able to turn the tide of low-carbon investments and reach investment shares that would prevail for a high-stranding projection.

Finally, for a benchmark return rate  $R^*$  equal to zero, the limit is not equal to 0.5, unlike for the case  $\Gamma = 0$ . This is because, for  $\Gamma$  very high or close to infinity, the censoring of opinions becomes more relevant for aggregate investment behaviour. As a result, unless  $-R_0 = 2(R_1 - R_0) \iff R_0 = -R_1$ , *i.e.* a case of perfect symmetry in censoring, high belief heterogeneity will introduce a distortion if beliefs are censored. Even in a situation of indifference  $(R_0 = 0)$ , the aggregate investment behaviour will be biased. For our calibration,  $\frac{-R_0}{R_1 - R_0} \approx \frac{3.43}{6.59+3.43} = 0.3426$ , which denotes a bias in disfavour of low-carbon technology.



**Fig. 8.** Interaction between long-run belief heterogeneity  $\bar{\sigma}$  and the discount rate  $\rho$ . Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.995" on Panel (a) corresponds to all  $(\bar{\sigma}, \rho)$  combinations for which the share of low-carbon investment  $\ell_I$  is equal to 0.995 in the case of high stranding central expectation.

#### 4.3. The discount rate and belief heterogeneity

We now explore the effects of the discount rate  $\rho$  on investment behaviours and how it interacts with belief heterogeneity.<sup>16</sup> While the strong dependence on other parameters does not allow us to derive analytical results for the effect of discount rate  $\rho$  on aggregate investments, it is possible to draw some insights from numerical simulations.

As Fig. 8 shows, for high-stranding plans (Panel (a)),  $\rho$  and  $\bar{\sigma}$  interact non-linearly. For sufficiently low levels of belief heterogeneity, the effect of the discount rate is negligible. In other words, when  $\bar{\sigma}$  is low, an increase in  $\rho$  requires only a marginal change in  $\bar{\sigma}$  to remain on the same isovalue curve. For higher levels of belief heterogeneity, instead, the relationship become strongly nonlinear. When discount rate  $\rho$  is low, remaining on the same isovalue line requires a higher belief heterogeneity, *i.e.*, an equal level of belief uncertainty would yield a higher low-carbon investment share. When  $\rho$  becomes higher, its effect on low-carbon investments becomes unambiguously negative, as lower belief heterogeneity is now necessary to compensate for the increase in  $\rho$  and remain on the same isovalue line. This pattern emerges because of two opposite effects of the discount rate. On the one hand, the discount rate decreases the weight of periods over which agents expect a higher payoff for low-carbon energy in the central expectation. On the other hand, because longer-run beliefs are also discounted, agents tend to give less weight to opinions relevant for later periods. This suggests that the belief-discounting effect of  $\rho$  dominates the discounting of the central expectation for low values of  $\rho$ . The opposite is true for higher values of  $\rho$ . It also explains why belief heterogeneity and  $\rho$  should interact positively for high values of  $\rho$ . In the presence of high belief heterogeneity, discounting more those beliefs straying from the central expectation will tend to give even more weight to the central expectation, which yields less low-carbon investment due to higher discounting.

The low-stranding projection (Panel (b)), by contrast, entails an unambiguously negative impact of the discount rate on lowcarbon investment shares. We can see that the interaction between belief heterogeneity and discounting is reversed compared to the high-stranding case. The mechanism is the same as in the above: discounting heterogeneity decreases the weight of norm-contrarian beliefs for investor behaviours which, with a low-stranding central expectation, will reduce low-carbon investment.

This interaction between belief heterogeneity and the discount rate shows that, regardless of their beliefs, agents exhibiting a higher discount rate will invest less in low-carbon technology. It suggests that anchoring expectations and striving to decrease socially inadequate preferences for the present (Steffen, 2020) should go hand in hand. For low-stranding projections, an optimal discount rate is zero. For high-stranding ones, our results show that a non-zero discount rate maximises low-carbon investment through the lower weighting of heterogeneous beliefs over later periods.

#### 4.4. Farsightedness and belief heterogeneity

We now turn to the interactions between belief heterogeneity and farsightedness, denoted by the length of the planning horizon *S*. The general impact of increasing the planning horizon by one period is given by Proposition 8.

**Proposition 8.** Let us define  $R_S^*$  the value of the benchmark return rate for a given length for the planning horizon *S*. An increase in the planning horizon *S* has a positive effect if  $R_{S-1}^* + \pi_{L,S} - \pi_{H,S} > 0$ . Based on Proposition 1, there exists an  $s_1 \in [|1, S|]$  sufficiently large such that this condition holds (if  $\pi_{L,1} - \pi_{H,1} > 0$ ,  $s_1 = 1$ ). Still based on Proposition 1, there exists an  $s_2 > s_1$  such that the condition reverses if  $\pi_{L,1} - \pi_{H,1}$  is negative and low enough.  $s_2$  increase with  $\ell^{\tilde{e}}$  and  $b_{\ell}$ .

**Proof.** Demonstration is given in Appendix F.8.

<sup>&</sup>lt;sup>16</sup> For the sake of brevity, we only allow  $\bar{\sigma}$  to vary in modulating belief heterogeneity in our illustrations. Effects would be qualitatively similar if we changed the level of  $\sigma_0$ , as long as we keep  $\sigma_0 < \bar{\sigma}$ .



**Fig. 9.** Interaction between long-run belief heterogeneity  $\bar{\sigma}$  and the planning horizon *S*. Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.5" on Panel (a) corresponds to all ( $\bar{\sigma}$ , *S*) combinations for which the share of low-carbon investment  $\ell_I$  is equal to 0.5 in the case of high stranding central expectation.

Intuitively, agents with a longer planning horizon account for interval tend to include more periods *s* where low-carbon tech is expected to be more profitable. However, as shown in Proposition 8, this effect is compensated for by negative  $\pi_{L,s} - \pi_{H,s}$  emerging once the stranding period ends. This latter effect is exacerbated by low discounting.

These figures illustrate the fact that a minimum planning horizon is required for low-carbon investment to emerge in the current period. This minimum planning horizon corresponds to the expectation horizon necessary for agents to account for enough *s*-time intervals so to tilt the expected profit rate in favour of low-carbon technologies.

Starting with the high-stranding central expectation (Panel (a)), increasing belief heterogeneity has two opposite effects. It increases low-carbon investment for low planning horizons, consistently with the discussion above. It can also decrease by a moderate amount ( $\approx 10\%$ ) the share of low-carbon investment if the horizon of expectation is long and belief heterogeneity is high.<sup>17</sup> This feature is due to our assumption of time *s*-increasing opinion diversity. As the planning horizon increases, expectations will diverge more for later periods. Hence, agents will tend to exhibit a more balanced investment portfolio as their planning horizon increases. This highlights a trade-off between sufficient long-termism and the uncertainty linked to very late periods. Finally an increase in belief heterogeneity decreases the cutoff point from which the effect of a longer planning horizon entails lower low-carbon investment. This is because higher heterogeneity will tend to decrease the expected value of positive  $\pi_{L,s} - \pi_{H,s}$  over  $[|s_1, s_2|]$  in aggregate, resulting in negative  $\varphi^*$  if the planning horizon is sufficiently longer to include periods over which agents expect low-carbon technologies to be less profitable.

The low-stranding case (Panel (b)) shows a synergy between the length of the planning horizon and belief heterogeneity. It is because, with higher belief dispersion, including more periods into the planning horizon will lead a greater part of the investors' population believe that low-carbon investments will be profitable, and for longer. Conversely, if belief heterogeneity is low, increasing the planning horizon has only a very moderate impact low-carbon investment. It is because agents expect that the period of stranding over which low-carbon investments will be more profitable is only transitory. Because the central projection features low stranding, the stranding period will be short and too transitory to outweigh subsequent periods, over which the edge of the low-carbon technology is reduced. Quantitative implications are greater than in the high-stranding case and function of the length of the planning horizon. Intuitively, for a planning horizon maximising stranding expectation, allowing for greater belief heterogeneity pushes even farther the range of transition-prone expectations. Hence, the share of low-carbon investors will increase.

These results also show that a relatively high share of low-carbon investment (40-50%) can emerge for some planning horizons in the event of important belief heterogeneity, regardless of the underlying central expectation. As sketched in Sections 4.2 and 4.3, introducing belief heterogeneity introduces a whole gradient of interior solutions between 0 and 1. A same share of low-carbon investment can correspond to many parameter constellations. As a result, considering observed shares of low-carbon investment alone as an indicator of the good health of the transition can be misleading.

For instance, the share of renewable investment in Europe revolves around 40%, which may be seen as encouraging. Yet, policymakers should be cautious, in that this figure may correspond equally to a state of high central expectation, but short planning horizon (Fig. 9 Panel (a) or, conversely, to a state of long planning horizon, but high belief dispersion (Panel (b)). Given that both possibilities lead to different policy implications, regulators should consider surveying investors to gauge as much as possible to state of the market wisdom, how dispersed beliefs are, and how short-sighted agents are.

#### 5. Conclusion

Transitioning to a carbon-free economy requires convincing a sufficiently relevant proportion of private decision-makers that investing in low-carbon technology is the most profitable strategy. High-carbon technologies still often represent the most convenient investment alternative due to higher productivity, better financing conditions, and other factors. However, firms might decide to invest in low-carbon technologies if they expect them to diffuse rapidly and, consequently, high-carbon capital units to become under-utilised ('stranded') and deliver lower profits in the future.

<sup>&</sup>lt;sup>17</sup> This feature shows in Panel (a) through the upward-sloping shape of the "0.9" line.

Firms have diverse beliefs about what will happen in the future. We introduce heterogeneity by representing firms' opinions as normally distributed around a central expectation, which we take as an announced policy pathway or a common market norm, and increasing logistically in the length of their planning horizon. Our model allows us to explore the role of opinion diversity on investment behaviours, relaxing the hypothesis of coordinated expectations usually encountered in macroeconomic modelling.

Our analytical framework is not immune to limitations, paving the way for future research. The assumption of homogeneous time preferences could be relaxed by representing populations with heterogeneous planning horizons and discount rates. Including expectations for other variables than the utilisation rate, as we start exploring in Appendix B, would be equally relevant, possibly down to the conception of a full 'mental model' of the economy for the agents, close to Gabaix's (2014) endeavour. Our work could also include different and more complex distributions for opinion diversity, as suggested by Appendix A. Finally, exploring the dynamic implications of the model is likely to offer valuable insights. For instance, competing market norms across which agents could switch (in a way similar to Franke and Westerhoff, 2017; Hommes, 2021, among others) could be included; or climate policies and their implications could be explicitly represented. However, despite its limitations, we believe our paper offers a novel perspective on the low-carbon transition dynamics, with a framework that is close enough to reality to capture several key behavioural dimensions for the first time, while maintaining analytical tractability.

We find that the effects of belief heterogeneity on investment shares depend significantly on the existing market norm. If central market expectations entail a rapid transition and large high-carbon stranding, stronger belief heterogeneity reduce low-carbon investment. The opposite is true for central expectations implying low stranding expectations. The strength of this effect is strongly non-linear: in a context of both very polarised or fully balanced market expectations, investment shares will not react much to changes in belief heterogeneity; by contrast, milder central expectations in favour of either technology will be very sensitive to changes in heterogeneity. Another key finding is that high-stranding market norms will deliver higher low-carbon investment shares, even with high belief dispersion. Our main policy takeaway is that policymakers should strive to estimate the state of existing central expectations and of belief heterogeneity in the marketplace in order to best anticipate possible market reactions to real-world developments. They could also try to infuse high-stranding market norms within the business community, if their goal is to increase the share of low-carbon energy, with the caveat that such policy move should not endanger their credibility.

We also find that under our benchmark calibration, and for a sizeable proportion of the parameter constellation we explore, low-carbon investment shares are more sensitive to shocks to short-run belief heterogeneity than to long-run belief heterogeneity. For moderate overall belief dispersion, this implies that shocks to short-run belief dispersion can have disproportionate effects. In this respect, we suggest that policymakers should make sure that agents' expectations are aligned as much possible regarding short-run outcomes. This could be done, for instance, by adopting short-term planning involving all stakeholders to reach consensus, with regular evaluation from an independent authority. These findings finally suggest that policymakers should assess investors' views about the low-carbon transition — e.g. through surveys or expert auditions — to fully grasp whether the dynamics of today will keep going in the future.

Finally, we study how heterogeneous beliefs interact with agents' time preferences along two dimensions. Belief heterogeneity interacts with the discount rate in various ways. If the central expectation has high stranding content, higher belief heterogeneity exacerbates the effects of the discount rate. Reciprocally, higher belief heterogeneity tones down the impact of the discount rate if the central expectation has low stranding content. In that case, transition believers tend to expect more stranding in the early periods of their planning horizon. More importantly, we show that belief heterogeneity can thwart the positive effects of longer planning horizons. Indeed, while increasing the planning horizon augments expected stranding, it also increases the impact of opinion diversity by widening the range of projections. Again, the final result depends on the underlying central expectations. In particular, belief heterogeneity can have a highly detrimental effect if the central expectation has low stranding content by keeping low-carbon investment close to a 50-50 dispatch. Finally, we find that, in the presence of belief heterogeneity, simply looking at the share of low-carbon investment to assess the state of investor expectations is misleading, as the same share of low-carbon investment can correspond to many different states of investor opinion, lengths of the planning horizon, or levels of the market norm.

## Declaration of competing interest

The authors declare the following financial interests/personal relationships which may be considered as potential competing interests: Emanuele Campiglio reports financial support was provided by European Research Council.

#### Data availability

Data will be made available on request.

#### Appendix A. Using a stable distribution for belief heterogeneity

We present here some insights on the use of alternative distributions to depict belief heterogeneity. The normal distribution has the important drawback of being symmetric, which does not allow for skewed distributions. It also puts great emphasis on values around the mean. The normal distribution, however, is a special case of a broader family of distributions, Stable laws. They are stable by addition and whose skewness and kurtosis can be parameterised.

Stable laws are not directly depicted by a density, but can be derived from their characteristic function. With  $X \sim Stable(\alpha, \gamma, \beta, \delta)$ , the characteristic function writes, with *i* the imaginary number and  $\forall t \in \mathbb{R}$ :



Fig. A1. Comparison of the Normal distribution benchmark to various Stable laws for the same  $\sigma_{us}$  schedule.

$$\mathbb{E}(e^{itX}) = \begin{cases} \exp\left(-\gamma^{\alpha}|t|^{\alpha} \left[1 + i\beta sign(t) \tan(\frac{\pi\alpha}{2})((|\gamma|)^{1-\alpha} - 1)\right] + i\delta t\right) & \text{if } \alpha \neq 1\\ \exp\left(-\gamma|t| \left[1 + i\beta sign(t)\frac{2}{\pi}\ln(\gamma|t|)\right] + i\delta t\right) & \text{if } \alpha = 1 \end{cases}$$

This function is parameterised by:

- $\alpha \in [0; 2]$ , named the 'stability parameter', which is a proxy for kurtosis.  $\alpha = 2$  is a normal distribution.
- $\gamma \in \mathbb{R}^+$  is the scale parameter, which rules the range on which the distribution will take most of its values and is therefore a proxy for the variance.
- $\beta \in [-1;1]$  is the skewness parameter.  $\beta = 1$  indicates rightward skewness and vice-versa.
- $\delta \in \mathbb{R}$  is a position parameter, which is an approximation for the mode for high  $\alpha$ s.

The stable distribution is called  $\alpha$ -stable, in the sense that the sum of stable distributions with the same  $\alpha$ s is a stable distribution. More precisely, with  $X_1 \sim Stable(\alpha, \gamma_1, \beta_1, \delta_1)$  and  $X_2 \sim Stable(\alpha, \gamma_2, \beta_2, \delta_2)$ , the sum  $X_1 + X_2 \sim Stable(\alpha, \gamma, \beta, \delta)$ , with:

$$\gamma = (\gamma_1^{\alpha} + \gamma_2^{\alpha})^{\frac{1}{\alpha}}$$
$$\beta = \frac{\beta_1 \gamma_1^{\alpha} + \beta_2 \gamma_2^{\alpha}}{\gamma_1^{\alpha} + \gamma_2^{\alpha}}$$
$$\delta = \delta_1 + \delta_2$$

This can be easily generalised to the sum of *n* stable distributions, which allows us to define our  $\varepsilon_{u,s}$  much more generally than with a normal distribution while still being able to compute our aggregate profitability metric  $R^*$ .

However, worth emphasising is that stable distributions are in general not as readily interpretable as the Gaussian special case. Typically, the variance is undefined for  $\alpha < 2$  (*i.e.* any case that is not a normal distribution) and the mean value is undefined for  $\alpha < 1$ . As a result, the two parameters  $\gamma$  and  $\delta$  are only proxies for respectively the variance and the mode of the distribution in the general case. Any result should therefore be taken with precaution given their lesser interpretability.

We nonetheless parameterise our  $\varepsilon_{u,s}$ , such that  $\delta_{u,s} = 0$  and  $\gamma_{u,s} = \sigma_{u,s}$ , leaving  $\beta$  and  $\alpha$  free. To keep an interpretable 0-mean, we modulate  $\alpha$  to keep it within the ]1,2[ interval. To study the impact of leftward-skewed beliefs (*i.e.*, a greater mass of agents believing in stranding), we set to start with  $\beta = -1$ . We illustrate how this constellation of parameters changes the distribution with respect to a Normal benchmark in Fig. A1, for a same scale ( $\sigma_{u,s}$ ) schedule and various  $\alpha$ s.

As can be seen, distributions are obviously more skewed leftward. Plus, the mode tends to shift away from the zero mean to make for the skewness. Finally, non-normal distributions tend to de-emphasise values around the mode and focus on 'rarer' events to the left. However, as can be seen, they also tend to include more events to the right of the zero-mean. As a result, especially for low  $\alpha$ s, Stable distributions are useful in representing populations that are more polarised. We display results in Fig. A2 for the  $\sigma_0$  and  $\bar{\sigma}$ , over a smaller compact than in Section 4 to make it more tractable and for various values of  $\alpha$ .

As can be seen, changing the distribution has both qualitative and quantitative implications. Interestingly, more skewness leads most often to lower low-carbon investment, even if it makes a greater part of the population believing in high stranding. This is entirely attributable to the fact that skewness is compensated by a greater share of the population believing in higher stranding.



Fig. A2. Sensitivity figures on dissent parameters  $\sigma_0$  and  $\bar{\sigma}$ .

Because of our censoring process, this share of the population will become more relevant as the central stranding projection hits lower values, resulting in lower low-carbon investment values. Interestingly, however, this logic gets reversed for low  $\alpha$ s and high  $\sigma_{u,s}$  schedules, as the population believing in high stranding is more important. These results suggest that a greater polarisation of beliefs acts in disfavour of the low-carbon transition, unless it is characterised by a very 'strong minority'. Note nonetheless that, within our parameter range, the impact of this strong minority seems reduced.

# Appendix B. Expanding the realm of uncertainty

In this Appendix, we offer some insights on how our results would change if belief heterogeneity went beyond the utilisation rate of high-carbon capital to also affect the prices of both electricity and fossil fuels. We redefine the return rates for the two technologies as:

$$r_{H} = \sum_{s=t}^{S} \beta^{s} \left[ \left( p_{E} - \frac{p_{F}}{\xi_{F}} + \varepsilon_{\pi,s}^{H} \right) \xi_{H} (u_{H,s}^{*} + \varepsilon_{u,s}) - \alpha_{H} \psi_{H} c_{H} \right]$$

$$r_{L} = \sum_{s=t}^{S} \beta^{s} \left[ (p_{E} + \varepsilon_{\pi,s}^{L}) u_{L}^{f} \xi_{L} - \alpha_{L} c_{L} \right]$$

$$(26)$$

$$(27)$$

where  $\epsilon_{\pi,s}^{(H,L)} \sim \mathcal{N}\left(0, (\sigma_{\pi,s}^{(H,L)})^2\right)$  is now the dispersion of beliefs around the normal expected unit profitability for both technologies. These terms include movements in intermediate fossil input prices  $p_F$  (including taxes) and changes in electricity prices  $p_E$ .

By recalling our investment rule for which  $\ell_I = P(r_L - r_H > 0)$  and developing the expression, we can write:

$$\mathcal{E}_{I} = P\left(\underbrace{\sum_{s=t}^{S} \beta^{s} \left[ \left( p_{E} - \frac{p_{F}}{\xi_{F}} \right) \xi_{H} u_{H,s}^{*} - p_{E} \xi_{L} u_{L}^{f} - \alpha_{H} \psi_{H} c_{H} + \alpha_{L} c_{L} \right]}_{"Mean \ belief" term} + \underbrace{\sum_{s=t}^{S} \beta^{s} \left[ \epsilon_{u,s} \xi_{H} \left( p_{E} - \frac{p_{F}}{\xi_{F}} \right) + \epsilon_{\pi,s}^{H} \xi_{H} u_{H,s}^{*} - \epsilon_{\pi,s}^{L} \xi_{L} \right]}_{Normal \ distribution \ term} + \underbrace{\sum_{s=t}^{S} \beta^{s} \left[ \epsilon_{u,s} \epsilon_{\pi,s}^{H} \xi_{H} \right]}_{S=t} \ge 0$$

(28)

Equation (28) yields a product of Normal random variables (bottom term in the equation), which is known to be described by a Normal-Product distribution. The sum of such a distribution, which we need to derive our aggregate return rate, does not have a readily available functional form. However, if the random variables are assumed to be independent, the central-limit theorem allows us to approximate the whole distribution by a well-parametrised Normal distribution. Ware and Lad (2003) further show that for Normal-Product distribution, this approximation holds once we sum as few as five products of normal random variables. Hence, because we sum more than five random variables in our benchmark case with S = 20, all the random variables having a mean of zero, and since we assume independence between  $\varepsilon_{u,s}$  and  $\varepsilon_{\pi,s}^H$ , products can be well approximated by a Normal distribution with

mean zero and variance  $\left(\epsilon_{u,s}\epsilon_{\pi,s}^{H}\xi_{H}\right)^{2}$ .

As a result, waiving the censoring described in Appendix E for simplicity, we can write:

$$\ell_{I} \sim \mathcal{N}\left(\underbrace{\sum_{s=t}^{S} \beta^{s} \left[ \left( p_{E} - \frac{p_{F}}{\xi_{F}} \right) \xi_{H} u_{H,s}^{*} - p_{E} \xi_{L} u_{L}^{f} - \alpha_{H} \psi_{H} c_{H} + \alpha_{L} c_{L} \right]}_{Mean}, \left( \underbrace{\sum_{s=t}^{S} \beta^{2s} \left( \sigma_{u,s}^{2} \xi_{H} \left( p_{E} - \frac{p_{F}}{\xi_{F}} \right)^{2} + \left( \sigma_{\pi,s}^{H} \xi_{H} u_{H,s}^{*} \right)^{2} + \left( \sigma_{\pi,s}^{L} \xi_{L} u_{L}^{f} \right)^{2} + \left( \sigma_{u,s} \sigma_{\pi,s} \xi_{H} \right)^{2} \right)}_{Standard - Error} \right)$$

$$(29)$$

Hence, the Normal distribution above allows us to maintain the same Probit approach we use in the main text. Equation (29) is a direct extension of our basic formula described in Equations (11)-(16) and with three additional terms:

- $\left(\sigma_{\pi,s}^{H}\xi_{H}u_{H,s}^{*}\right)^{2}$ , the belief dispersion on the profitability of high-carbon technology;
- $\left(\sigma_{\pi,s}^{L}\xi_{L}u_{I}^{f}\right)^{2}$ , the belief dispersion on low-carbon profitability;
- $(\sigma_{\mu_s}\sigma_{\pi_s}\xi_H)^2$ , the interaction between belief dispersion on the profitability of high-carbon technology and on the utilisation rate.

Notice that expanding the realm of uncertainty has, quantitatively, the same implications as shifting the  $\sigma_{u,s}$  schedule in the case when there is only belief dispersion around future utilisation rates (asset stranding). We therefore expect this augmented model to yield lower values of  $\ell_I$ , all other things left equal. Furthermore, increasing any  $\sigma$  would yield the same qualitative results as in the benchmark case with beliefs only on future utilisation rates.

Without providing analytical proofs for brevity, we report some results along the line of Section 4 by supposing that belief dispersion on  $\pi_L$  and  $\pi_H$  ( $\sigma_{\pi_L}$  and  $\sigma_{\pi_H}$ ) increases through time *s* following the same logistic behaviour identified for  $\sigma_{u_H}$  and with the same default minimum and maximum belief dispersion (*i.e.*  $\sigma_{\pi_H,0} = \sigma_{\pi_H,0} = \sigma_{u_H,0} = 0.01$  and  $\bar{\sigma}_{\pi_L,0} = \bar{\sigma}_{u_H,0} = 0.5$ ).

First, we study how the introduction of these new dimensions of belief uncertainty affects our results. For this purpose, we reproduce Fig. 6, including our new hypotheses on  $\sigma_{\pi_L}$  and  $\sigma_{\pi_H}$ . Results are reported in Fig. A3. For the ambitious scenario (high stranding), results are close to those in Fig. 6, qualitatively and quantitatively. However, effects are more in disfavour of low-carbon energy sources overall: a full low-carbon investment share is never reached, with a 95% maximum, and the lower values are reached for high belief dispersion. Furthermore, the effects of  $\bar{\sigma}_{u_H}$  and  $\sigma_{u_{H,0}}$  are much more non-linear, with sharper curvatures. This is



**Fig. A3.** Long-run and short-run belief heterogeneity (extended uncertainty). Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.85" on Panel (a) corresponds to all  $(\bar{\sigma}_{u_H}, \sigma_{u_H,0})$  combinations for which the share of low-carbon investment  $\ell_I$  is equal to 0.85 in the case of high stranding central expectation.



**Fig. A4.** Long-run belief heterogeneity on low- and high-carbon profitability (extended uncertainty). Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.85" on Panel (a) corresponds to all  $(\bar{\sigma}_{\pi_L}, \bar{\sigma}_{\pi_H})$  combinations for which the share of low-carbon investment  $\ell_I$  is equal to 0.85 in the case of high stranding central expectation.

attributable to: (i) belief dispersion on low-carbon energy profitability, which tends to decrease low-carbon investment; and (ii) the interaction term  $(\sigma_{u_H,s}\sigma_{\pi_H,s}\xi_H)^2$ , which exacerbates the effects of uncertainty on asset stranding. In the unambitious case (low stranding), we find again effects to be in disfavour of low-carbon energy sources compared to our default case, due to the belief dispersion on low-carbon energy profitability. We also find a non-linear effect, with  $\bar{\sigma}_{u_H}$  and  $\sigma_{u_H,0}$  associated with *less* low-carbon investment than lower combinations, especially for high  $\sigma_{u,0}$ . This non-linearity emerges because of the interaction term  $\sigma_{\pi_H}\sigma_{u_H}$ , which amplifies the effect of high  $\bar{\sigma}_{u_H}$  and  $\sigma_{u_H,0}$  in late periods. In this context, agents hardly take into account the slow progresses of low-carbon energy in the long run prevailing under the central expectation, and consider, in the aggregate, low-carbon energy to be overall less profitable than for lower levels of belief dispersion. Hence a lower share of low-carbon energy for high belief dispersion levels in the unambitious case. Otherwise, results are qualitatively similar, albeit sharper due to the interaction between  $\sigma_{u_H}$  and  $\sigma_{\pi_H,s}$ .

Second, we fix belief dispersion parameters on the utilisation rate to their benchmark values and modulate the maximum value of  $\sigma_{\pi_L}$  and  $\sigma_{\pi_H}$ . As above, we differentiate between an ambitious and an unambitious central expectation. Results are displayed in Fig. A4. The effects of higher belief dispersion on profitability are less pronounced than the ones associated to capacity utilisation belief dispersion, for both scenarios. This is because these beliefs apply to variables that do not move in time *s*, while we assume a moving central projection for stranding. However, we never reach a full investment in low-carbon technologies, the highest achieved value being 95% in the ambitious scenarios. We also find that the effects of both types of belief dispersion are non-linear, but not symmetrical. In the high-stranding scenarios, long-run belief heterogeneity on  $\sigma_{\pi_L}$  has relatively low effect on investment behaviour for low values, as figured by the flatness of the upper part of the curves in the bottom-left corner. These magnitudes are very close to the effects of  $\sigma_{u_H}$  shown in Fig. 6. Conversely,  $\sigma_{\pi_H}$  exhibits much larger effects. This is explained by the multiplicative term including  $\sigma_{\pi_H}$  and  $\sigma_{u_H}$ , which exacerbates the effects of any increase in  $\sigma_{\pi_H}$ . This, however, only holds true until a certain point, after which effects become relatively linear. Results are qualitatively similar, but reversed, in the non-ambitious case.

All in all, extending the realm of uncertainty bears similar qualitative results to the analysis provided in the main text. Quantitatively, it tends to reduce low-carbon investment shares due to uncertainty on future profitability. Most importantly, the effects of belief dispersion are sharper and more non-linear due to interactions between different belief items. This *compound* effect of belief dispersion suggests that several types of uncertainties can exacerbate each other and that regulators should aim to tackle various types of uncertainties simultaneously as much as possible.



**Fig. A5.** Long-run and short-run belief heterogeneity (hydro in low-carbon category). Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.85" on Panel (a) corresponds to all  $(\bar{\sigma}, \sigma_0)$  combinations for which the share of low-carbon investment  $\ell_i$  is equal to 0.85 in the case of high stranding central expectation.



**Fig. A6.** Long-run and short-run belief heterogeneity (hydro and nuclear energy in low-carbon category). Panel (a) describes our high-stranding case, while Panel (b) shows our low-stranding case. The charts plot isovalue lines. For instance, the curve with legend "0.85" on Panel (a) corresponds to all ( $\bar{\sigma}, \sigma_0$ ) combinations for which the share of low-carbon investment  $\ell_I$  is equal to 0.85 in the case of high stranding central expectation.

#### Appendix C. Alternative dispatch composition for low- and high-carbon technologies

In this Appendix, we consider alternative dispatch compositions for the definition of our 'low-carbon' category to address potentially misleading results linked to our choice of including hydropower and nuclear technologies in the 'high-carbon' category.

First, we include hydro in the low-carbon category. This brings the initial share of low-carbon energy production to 34%, and total low-carbon capacity to 438 GW, *i.e.* around 45% of total installed capacity (Eurostat, 2021). Productivity parameters are redefined accordingly to  $\xi_H \approx 5.0221$  and  $\xi_L \approx 2.8745$ . To adjust for the change in the starting value of the low-carbon share, we recalibrate the intrinsic growth rate of our ambitious and unambitious scenarios to 0.2 and 0.1, respectively, while setting the long-term share to 95% after 30 years, consistently with the goals laid in the Fit for 55 package (EC, 2021). We reproduce here results displayed in Fig. 6, to compare the effect of short- and long-run belief heterogeneity. As shown in Fig. A5, results are qualitatively very close to the ones obtained with our default dispatch. The dichotomy between ambitious and unambitious scenarios remains, and we still find the decreasing marginal effect of increasing uncertainty in both cases. Our effects become slightly sharper than in the benchmark case due to the redefinition of our ambitious and unambitious scenarios, which tend to favour the incumbent.

Second, we also introduce nuclear power. Given the high weight of this energy source in the European mix, including nuclear brings the low-carbon share of energy to 53%, and total low-carbon capacity to around 550 GW, *i.e.* around 55% of total installed capacity Eurostat (2021). Productivity parameters are redefined as  $\xi_H = 4.4942$  and  $\xi_L = 3.69$ . Given the large change to our energy shares, we redefine our ambitious and unambitious scenarios. The intrinsic growth rate  $b_\ell$  for the ambitious scenario is decreased from 0.25 to 0.1; the carrying capacity  $\ell^{\bar{\ell}}$  is moved from 0.9 to 0.95 as in the case above. For the unambitious scenario, while retaining the same carrying capacity of 0.95, we assume an intrinsic growth rate of 0.05. Results are displayed in Fig. A6.

In the ambitious scenario (high stranding), results are similar to the ones with the default dispatch composition, with similar gradients and effects. For the unambitious scenario (low stranding), the model yields a non-linearity in belief dispersion for low values, as shown by the outward orange line. Further, the overall pattern matches that of the ambitious scenario, with most cold-coloured line corresponding to low levels of low-carbon investments. Results differ due to the redefinition of productivity parameters. Although they still give an edge to the high-carbon sector for low levels of belief heterogeneity, this edge is very small due to the high productivity of nuclear power plants. As a result, for high level of belief dispersion, this edge is reversed (the spread between  $r_L$  and  $r_H$  becomes positive). This yields a behaviour similar to the ambitious scenario for high levels of belief dispersion. This result shows that, if the technologies are very close in terms of mean expected returns, non-linearities can emerge in the model. And, under certain conditions, such as those in the example above, belief uncertainty can diminish low-carbon investment even for otherwise unambitious scenarios.



Fig. A7. Shifting bounds along with a change in variable.

## Appendix D. Derivation of return rates

Heterogeneity in transition expectations creates heterogeneity in the expected return rates for the two technologies. Given equations (4) and (6), we can rewrite equation (1) for the high-carbon sector as

$$r_{H,t} = \sum_{s=t}^{S} \beta^{s} \left[ \left( p_E - \frac{p_F}{\xi_F} \right) \xi_H(u_{H,s}^* + \varepsilon_{u,s}) - \alpha_H \psi_H c_H \right].$$
(30)

Similarly to what was done with the expected capacity utilisation  $u_H$ , the expected return rate on a unit of  $K_H$  can be disaggregated into a 'rational stranding' deterministic part and an error term. Defining for convenience a new variable  $\gamma_H = (p_E - \frac{p_E}{\xi_F})\xi_H$ , equation (30) becomes  $r_{H,t} = R_H^* + \epsilon_r$ , with  $R_H^* = \sum_{s=t}^S \beta^s \left[ \gamma_H u_H^* - \alpha_H \psi_H c_H \right]$  and  $\epsilon_r \sim \mathcal{N}(0, \sum_{s=t}^S \beta^{2s} \gamma^2 \sigma_{u,s}^2)$ . The expected return rate on low-carbon capital  $r_H$ , on the other hand, lacks by assumption any random part. That is,  $r_L = R_L^* = \sum_{s=t}^S \beta^s \left[ p_E \xi_L - \alpha_L \psi_L c_L \right]$ . This gives us the net present value of future technological investments but it now remains to find an expression  $\vartheta$  such that Equation (14) is satisfied.

$$\Pr(r_L > r_H) = \Pr(r_L - r_H > 0) = \Pr(\vartheta > 0).$$
(31)

From the above we see that  $\vartheta = R_L^* - R_H^* - \varepsilon_r$ , but this expression must be scaled so the error term has a standard normal distribution. We thus divide the expression by the variance of  $\varepsilon_r$  to get our final expression  $\varphi$ :

$$\varphi = \frac{R_L^s - R_H^s}{\sum_{s=t}^s \beta^{2s} \gamma^2 \sigma_{u,s}^2} - \frac{\varepsilon_r}{\sum_{s=t}^s \beta^{2s} \gamma^2 \sigma_{u,s}^2} = \varphi^* - \varepsilon_\varphi, \tag{32}$$

where  $\varepsilon_{\varphi} \sim \mathcal{N}(0, 1)$ .

Furthermore, since the variance term  $\sum_{s=t}^{S} \beta^{2s} \gamma^2 \sigma_{u,s}^2 > 0$ ,

$$\Pr(r_L > r_H) = \Pr(\vartheta > 0) = \Pr(\frac{\vartheta}{\sum_{s=t}^{s} \beta^{2s} \gamma^2 \sigma_{u,s}^2 > 0} > 0) = \Pr(\varphi > 0).$$
(33)

As a final step, using the symmetry of the normal distribution,

$$\Pr(\varphi > 0) = \Pr(\varphi^* - \varepsilon_{\varphi} > 0) = \Pr(\varepsilon_{\varphi} - \varphi^* < 0) = \Pr(\varepsilon_{\varphi} < \varphi^*) = \Phi(\varphi^*).$$
(34)

### Appendix E. Censoring the bounds of the distribution

To deal with the caveat that the utilisation rate is clearly bounded, we have to find a way to factor in the clear technical constraints imposed on it. If  $u_H$  is between *a* and *b* we must find out what the bounds are for the error and associated latent variable and how we deal with it as a probability. By censoring the random part of  $u_H$  in the relevant bounds and shifting these bounds along with the variable, we get bounds for  $\epsilon_{\varphi}$ . The final probability is then calculated conditionally on these bounds.

We begin by censoring the normal random variables in our given bounds. Given a utilisation rate constrained in the interval ]a, b[, the error of  $u_{H,s}$ ,  $\varepsilon_{u,s}$ , is constrained by the inequality in Equation (18). Pictured in Fig. A7, we may shift the distribution down to centre it on 0, simply by subtracting the central value  $\mu_H$  from both bounds.

$$a - u_{H,s}^* < \varepsilon_{u,s} < b - u_{H,s}^*.$$

$$\tag{35}$$



Fig. A8. Representation of possible values for bounds A and B and a value of interest C.

Likewise, the random variable derived from  $\varphi$ ,  $\epsilon_{\varphi} = \frac{\sum_{s=t}^{S} \rho^{s} \gamma_{s} \epsilon_{u,s}}{\sum_{s=t}^{S} \rho^{2} \gamma^{2} \sigma_{u,s}^{2}}$ , is hence constrained by the following bounds:

$$\frac{\sum_{s=t}^{S} \beta^{s} \gamma_{s}(a - u_{H,s}^{*})}{\sum_{s=t}^{S} \beta^{2s} \gamma^{2} \sigma_{u,s}^{2}} < \varepsilon_{\varphi} < \frac{\sum_{s=t}^{S} \beta^{s} \gamma_{s}(b - u_{H,s}^{*})}{\sum_{s=t}^{S} \beta^{2s} \gamma^{2} \sigma_{u,s}^{2}}.$$
(36)

Let *A* and *B* be the lower and upper bounds of  $\varepsilon_{\varphi}$  in Equation (36), respectively and let  $C = \mu_{\varphi}$ . We proceed with the probability as in Equation (16), but this time it is calculated as a conditional probability using the censored standard normal variable that we have derived. The situation is pictured in Fig. A8, with the value of interest *C* between bounds *A* and *B*. The probability that  $\varepsilon_{\varphi}$  is less than *C* is the area left of the value C,  $\Phi(C) - \Phi(A)$ . We must account for the limited values of possibility, so we divide by the shaded area representing all possible values,  $\Phi(B) - \Phi(A)$ , giving us Equation (37).

$$\mathscr{E}_{I} = \Pr(\varepsilon_{\varphi} < C \mid A < \varepsilon_{\varphi} < B) = \frac{\Phi(C) - \Phi(A)}{\Phi(B) - \Phi(A)}$$
(37)

Extending this process, we must account for the fact that the value C may be taken out of the realm of possibility. Thus we make a further extension to say that if the value is less than the possible range the probability is 0 and if higher then it is 1.

$$\mathcal{\ell}_{i} = \Pr(\varepsilon_{\varphi} < C \mid A < \varepsilon_{\varphi} < B) = \frac{\delta_{AC}(\Phi(C) - \Phi(A)) - \delta_{BC}(\Phi(B) - \Phi(C))}{\Phi(B) - \Phi(A)}$$

$$\text{where } \delta_{ij} = \begin{cases} 1 & i < j \\ 0 & i \ge j \end{cases}.$$

$$(38)$$

# Appendix F. Proofs

#### F.1. Proof of Proposition 1

**Proposition.** For  $b_{\ell}$  large enough,  $u_{H,s}^*$  is non-monotonic in  $s \in [|t+1, S|]$  and reaches a minimum in  $s_{min} \in [|1, S|]$ .

**Proof.** The proposition above is equivalent to showing that there exists an interval  $[|v, s_{min}|] \subset [|t + 1, S|]$  such that  $\frac{u_{H,s+1}^*}{u_{H,s}^*} < 1 \quad \forall s \in [|t + 1, s_{min}|]$  and  $\frac{u_{H,s+1}^*}{u_{H,s}^*} > 1 \quad \forall s \notin [|v, s_{min}|]$ . Let us first notice that  $u_{H,s+1}^*$  is defined as:

$$u_{H,s+1}^{*} = \begin{cases} \frac{e_{H,s+1}}{(1-\delta)K_{s}\xi_{H}} & \text{if} \quad I_{H,s}^{*} = 0\\ \frac{e_{H,s+1}}{(1-\delta)K_{H,s}^{d}\xi_{H}} & \text{if} \quad I_{H,s}^{*} > 0 \end{cases}$$

Hence that  $\frac{u_{H,s+1}^*}{u_{H,s}^*}$  can take 4 possible values:

$$\frac{u_{H,s+1}^{*}}{u_{H,s}^{*}} = \begin{cases} \frac{\frac{e_{H,s+1}(1-\delta)K_{s-1}\xi_{H}}{(1-\delta)K_{s}\xi_{H}e_{H,s-1}} & \text{if} \quad I_{H,s}^{*} = 0 \text{ and } I_{H,s-1}^{*} = 0 \quad (1) \\ \frac{e_{H,s+1}K_{H,s-1}^{d}\xi_{H}}{(1-\delta)K_{s}\xi_{H}e_{H,s-1}} & \text{if} \quad I_{H,s}^{*} = 0 \text{ and } I_{H,s-1}^{*} > 0 \quad (2) \\ \frac{e_{H,s+1}(1-\delta)K_{H,s-1}\xi_{H}}{K_{s}\xi_{H}e_{H,s-1}} & \text{if} \quad I_{H,s}^{*} > 0 \text{ and } I_{H,s-1}^{*} = 0 \quad (3) \\ \frac{e_{H,s+1}K_{H,s-1}^{d}\xi_{H}}{K_{H,s}^{d}\xi_{H}e_{H,s-1}} & \text{if} \quad I_{H,s}^{*} > 0 \text{ and } I_{H,s-1}^{*} > 0 \quad (4) \end{cases}$$

We focus on cases (1) and (2), since cases (3) and (4) describe cases in which no stranding is expected and therefore on which  $u_{H,s}$  is constant and equal to  $u^f$ .

Let us first notice that the condition for  $I_{H,s}^* = 0 \forall s$  is:

$$\frac{u_{H,s-1}}{u^f} < \frac{(1-\delta)}{(1+g_{e_{H,s}})}$$
(39)

It is easy to show that:

$$(1 - b_{\ell})(1 + g_e) < (1 + g_{e_{H,s}}) < (1 + g_e)$$
(40)

And that the sequence  $\left((1 + g_{e_{H,s}})\right)_{s \in [[t,S]]}$  is increasing and converges towards  $(1 + g_e)$ .

Hence, it is possible to find a *b* large enough such that Condition (1) is fulfilled at a given  $\bar{s}$ . For instance, the condition for  $\bar{s} = t + 1$  supposing that  $u_{H,t} = u^f$  yields:

$$1 < \frac{(1-\delta)}{(1+g_{e_{H,\bar{s}}})} \Leftrightarrow \left(1+g_{e}\right) \left(1-b\left(\frac{\bar{\ell}-\ell_{E,t}}{(1-\ell_{E,t})\bar{\ell}}\right)\right) < (1-\delta) \Leftrightarrow b > \frac{1-\frac{(1-\delta)}{(1+g_{e})}}{\left(\frac{\bar{\ell}-\ell_{E,t}}{(1-\ell_{E,t})\bar{\ell}}\right)}$$
(41)

This condition can be generalised for any s > t since, in expectations, as long as  $I_{H,s}^* > 0$ ,  $u_{H,s} = u^f$ . The condition for  $I_{H,\bar{s}+1}^* = 0$  is:

$$\frac{u_{H,5}}{u^f} < \frac{(1-\delta)}{(1+g_{e_{H,5}})}$$
(42)

In that case, we can write  $K_{H,\bar{s}+1} = (1-\delta)K_{H,\bar{s}} = (1-\delta)^2 K_{H,\bar{s}-1} = (1-\delta)^2 \frac{e_{H,\bar{s}-1}}{u_{\bar{s}-1}\xi_H}$  and Condition (1) can be rewritten as:

$$\frac{u_{H,\bar{s}-1}}{u^f} < \frac{(1-\delta)^2}{(1+g_{e_{H,\bar{s}}})(1+g_{e_{H,\bar{s}+1}})}$$
(43)

Which can again be fulfilled for *b* large enough. More generally, the condition for  $I_{H,k}^* = 0, k > \bar{s}$  writes:

$$\frac{u_{H,\bar{s}}}{u^f} < \frac{(1-\delta)^{k-\bar{s}}}{\prod_{i=\bar{s}}^k (1+g_{e_{H,i}})}$$
(44)

As we saw above, the sequence  $\left((1 + g_{e_{H,s}})\right)_{s \in [[t,S]]}$  is increasing and converges towards  $(1 + g_e) > (1 - \delta)$ . Since  $g_e$  and  $\delta$  are positive, based on the intermediate values theorem, there exists a  $s^*$  for which:

$$(1 + g_{e_{H,s^*}}) < (1 - \delta) < (1 + g_{e_{H,s^*+1}})$$

Hence that, for  $k > s^*$ , we can write  $\frac{(1-\delta)^{k-\bar{s}}}{\prod_{i=\bar{s}}^{k}(1+g_{e_{H,i}})}$  as follows:

$$\frac{(1-\delta)^{k-\bar{s}}}{\prod_{i=\bar{s}}^{k}(1+g_{e_{H,i}})} = \frac{1}{\prod_{i=\bar{s}}^{s^*}(1+g_{e_{H,i}})} \frac{(1-\delta)^{k-\bar{s}}}{\prod_{i=s^*+1}^{k}(1+g_{e_{H,i}})}$$

For *S* large enough,  $k^* < S$ . This result shows that there exists an interval  $U \subset [|t, S|]$  on which  $I^*_{H,s} = 0$  and that, for *S* large enough,  $k^* < S$ . For  $s > k^*$ ,  $I^*_{H,s} > 0$ .

The right-hand term is decreasing in k and converges towards zero. As a result, there exists a  $k^*$  for which:

$$\frac{u_{H,\bar{s}}}{u^f} > \frac{(1-\delta)^{k^*-\bar{s}}}{\prod_{i=\bar{s}}^{k^*}(1+g_{e_{H,i}})}$$
(45)

Cases (1) and (2)  $I_{H,k}^* = 0$  and  $I_{H,k-1}^* = 0 \forall k \in U$  thus depict situations arising for *b* large enough.

Considering case (1) and simplifying the corresponding equation, we get:

$$\frac{u_{H,s+1}^*}{u_{H,s}^*} = \frac{e_{H,s+1}(1-\delta)K_{H,s-1}\xi_H}{(1-\delta)^2 K_{H,s-1}\xi_H e_{H,s}} = \frac{e_{H,s+1}}{(1-\delta)e_{H,s}}$$

Which is below 1 if and only if  $\frac{e_{H,s+1}}{e_{H,s}} = (1 + g_{e_{H,s}}) < (1 - \delta)$ , that is, the decrease rate in high-carbon energy demand is superior to the depreciation rate. Using the fact that  $((1 + g_{e_{H,s}}))_{s \in [|t,S|]}$  is increasing and converges towards  $(1 + g_e) > (1 - \delta)$  and using again the intermediate value theorem, it follows that, for  $s > s_{min} = s^*$ ,  $\frac{e_{H,s+1}}{e_{H,s}} > 1$  and < 1 otherwise.

Considering now case (2) and noticing that in this instance  $K_{H,s} = K_{H,s-1}^d$ , we get:

$$\frac{u_{H,s+1}^{*}}{u_{H,s}^{*}} = \frac{e_{H,s+1}K_{H,s}^{d}\xi_{H}}{(1-\delta)^{2}K_{H,s}^{d}\xi_{H}e_{H,s}} = \frac{e_{H,s+1}}{(1-\delta)e_{H,s}}$$

Which yields the same condition as above: for  $b_{\ell}$  large enough, the utilisation rate will decrease.

**Lemma 1.** For  $b_{\ell}$  large enough, there exists an interval  $T \subset [|t, S|]$  such that,  $\forall s \in T$ ,  $\pi_{L,s} - \pi_{H,s} > 0$ .

**Proof.** The condition for  $\pi_{L,s} - \pi_{H,s} > 0$  is:

$$u_{H,s} < \frac{1}{p_E - \frac{p_F}{f_F}} (p_E \xi_L - \alpha_K \psi_L c_L^k + \alpha_H \psi_H c_H^k)$$

As per Proposition 1, with  $b_{\ell}$  large enough, it is possible to define a subset  $U \subset [[t, S]]$  on which  $u_{H,s}$  is decreasing, with the sequence  $(u_{H,s})_{s \in [[t+1,S]]}$  hitting a minimum at the highest value of U:  $s_{min}$ . Now, we know that,  $\forall s \in [[t+1,S]]$ :

$$u_{H,s+1}^{*} = \frac{e_{s}^{d}(1+g_{E}) - e_{L,s}(1+g_{E}) \left[1 + b_{\ell} \left(1 - \frac{e_{L,s}}{\bar{\ell}e_{s}^{d}}\right)\right]}{\xi_{H}[(1-\delta)K_{H,s} + I_{H,s}^{*}]}$$
(46)

A fortiori,

$$u_{H,s_{min}}^{*} = \frac{e_{s_{min}-1}^{d}(1+g_{E}) - e_{L,s_{min}-1}(1+g_{E}) \left[1 + b_{\ell} \left(1 - \frac{e_{L,s_{min}-1}}{\bar{\ell}e_{s_{min}-1}^{d}}\right)\right]}{\xi_{H}[(1-\delta)K_{H,s_{min}-1} + I_{H,s_{min}-1}^{*}]}$$
(47)

Which is an obviously decreasing function of  $b_{\ell}$ . Then, for  $b_{\ell}$  large enough,  $u_{H,s_{min}}^*$  will fulfil the condition above. It is further possible to define an interval  $[|s_{min} - a; s_{min} + a|]$  on which this condition holds, again for  $b_{\ell}$  large enough.

# F.2. Proof of Proposition 2

**Proposition.** For  $\sigma_{u,s} = 0$   $\forall s \in [|1, S|]$ ,  $\ell_I$  tends towards a degenerate probability distribution function, whereby:

$$\ell_I = \begin{cases} 0 & if \ R^* < 0 \\ 0.5 & if \ R^* = 0 \\ 1 & if \ R^* > 0 \end{cases}$$
(48)

**Proof.** Starting from  $\mathcal{C}_I = \frac{\Phi(\varphi^*) - \Phi(\varphi_0)}{\Phi(\varphi_1) - \Phi(\varphi_0)}$  and noticing that  $\varphi_0 = \frac{R_0}{\Gamma} < 0$  and  $\varphi_1 = \frac{R_1}{\Gamma} > 0$  within our parameter space, we have:

$$\lim_{\Gamma \to 0^+} \Phi\left(\varphi_0\right) = \lim_{\Gamma \to 0^+} \Phi\left(\frac{R_0}{\Gamma}\right) = 0$$
$$\lim_{\Gamma \to 0^+} \Phi(\varphi_1) = \lim_{\Gamma \to 0^+} \Phi\left(\frac{R_1}{\Gamma}\right) = 1$$

For  $R^* = 0$ , the result flows from the definition of  $\Phi$ . For any  $\Gamma$ , we have:

$$\ell_{I} = \frac{\Phi(0) - \Phi(\varphi_{0})}{\Phi(\varphi_{1}) - \Phi(\varphi_{0})} = \frac{0.5 - \Phi(\varphi_{0})}{\Phi(\varphi_{1} - \Phi(\varphi_{0}))}$$

Which, given the limits above, has  $\Phi(0) = 0.5$  as limit for  $\Gamma \rightarrow 0$ .

L. Cahen-Fourot, E. Campiglio, L. Daumas et al.

All in all,

$$\lim_{\Gamma \to 0^+} \ell_I = \lim_{\Gamma \to 0^+} \Phi(\varphi^*) = \begin{cases} 0 & if \ R^* < 0 \\ 0.5 & if \ R^* = 0 \\ 1 & if \ R^* > 0 \end{cases}$$
(49)

# F.3. Proof of Proposition 3

Let us consider a logistic sequence  $x_n = x_{n-1} \left( 1 + b(1 - \frac{x_{n-1}}{K}) \right)$  where K is a carrying capacity and b an intrinsic growth rate. Let us then consider  $x_0$  the first term of this sequence. Then,  $\forall n | x_n < K$ :

$$\frac{\partial x_n}{\partial K} \ge 0$$
(50)
$$\frac{\partial x_n}{\partial b} \ge 0$$
(51)

$$\frac{x_n}{\partial b} \ge 0 \tag{51}$$

$$\frac{\partial x_n}{\partial x_0} \ge 0 \tag{52}$$

With the last proposition holding for  $x_0 < K$ .

**Proof.** The first property can be shown by recurrence. Considering  $\frac{\partial x_n}{\partial K}$ , it is obvious, with  $x_0$  given that  $\frac{\partial x_1}{\partial K} > 0$ . Supposing that  $\frac{\partial x_n}{\partial K} \ge 0$ , we can show that  $\frac{\partial x_{n+1}}{\partial K} \ge 0$ .

$$\begin{aligned} \frac{\partial x_{n+1}}{\partial K} &= \frac{\partial x_n}{\partial K} (1 + b(1 - \frac{x_n}{K})) - x_n b \left( \frac{\frac{\partial x_n}{\partial K} K - x_n}{K^2} \right) \\ &= \frac{\partial x_n}{\partial K} (1 + b(1 - \frac{x_n}{K})) - b \frac{x_n}{K}) + \frac{x_n}{K^2} \\ &= \frac{\partial x_n}{\partial K} (1 + b(1 - 2b \frac{x_n}{K})) + \frac{x_n}{K^2} \end{aligned}$$

The condition for this expression to be negative is:

$$1 + \frac{\partial x_n}{\partial K} \frac{K^2}{x_n^2} < b(\frac{x_n}{K} - 1)$$

 $1 + \frac{K^2}{x_n^2} \frac{\partial K}{\partial x_n}$  is positive as per the assumption  $\frac{\partial x_n}{\partial K} > 0$  and  $\frac{x_n^2}{K^2} > 0$ . Hence that the condition above cannot hold for  $x_n \le K$ . We consider in the following only constellations of parameters for which  $x_n \le K$ , without loss of generality for the purpose of the paper. Regarding b, we can also proceed by recurrence. Defining:

$$\begin{aligned} x_n &= x_{n-1}(1+b(1-\frac{x_{n-1}}{K})) \\ x'_n &= x'_{n-1}(1+(b+db)(1-\frac{x'_{n-1}}{K})) \\ &= x'_{n-1}(1+(b')(1-\frac{x'_{n-1}}{K}))x_0 = x'_0 \end{aligned}$$

It is easy to check that, for  $x_0$  given,  $x'_1 - x_1 \ge 0$ . Supposing that  $x'_n - x_n \ge 0$ , we write:

$$\begin{aligned} x'_{n+1} - x_{n+1} &= x'_n (1 + (b')(1 - \frac{x'_n}{K})) - x_n (1 + b(1 - \frac{x_n}{K})) \\ &= x'_n - x_n + b'(1 - \frac{x'_n}{K}) - b(1 - \frac{x_n}{K}) \\ &\geq x'_n - x_n + b'(1 - \frac{x'_n}{K}) - b'(1 - \frac{x_n}{K}) \\ &\geq x'_n - x_n + b'(\frac{x_n}{K} - \frac{x'_n}{K}) \\ &\geq (x'_n - x_n) \frac{(1 - b)}{K} \end{aligned}$$

Given that  $(1 - \frac{b}{K}) > 0$  for b < 1 and null when b = 1, we can consider this condition to hold for reasonable values of b. For  $x_0$ , we can use once again the same method. Defining:

$$x_n = x_{n-1}(1 + b(1 - \frac{x_{n-1}}{K})), \ x_0 = x_0$$

L. Cahen-Fourot, E. Campiglio, L. Daumas et al.

$$\begin{aligned} x_n' &= x_{n-1}'(1+b)(1-\frac{x_{n-1}'}{K})) \ , \ x_0 &= x_0' \\ x_0 &\leq x_0' \end{aligned}$$

Showing that  $x'_1 \ge x_1$  and supposing that  $x'_n \ge x_n$ , we get the similar condition:

$$\begin{aligned} x'_{n+1} - x_{n+1} &= x'_n (1+b)(1 - \frac{x'_n}{K})) - x_n (1 + b(1 - \frac{x_n}{K})) \\ &= x'_n - x_n + b'(1 - \frac{x'_n}{K}) - b(1 - \frac{x_n}{K}) \\ &= x'_n - x_n + b'(\frac{x_n}{K} - \frac{x'_n}{K}) \\ &= (x'_n - x_n) \frac{(1-b)}{K} \end{aligned}$$

Which again holds true for  $b \le 1$ .  $\Box$ 

**Lemma 2.** All else held equal, increasing  $\overline{\ell}$  or  $b_{\ell}$  will have a positive effect on low-carbon investment.

**Proof.** This property follows from Proposition 2,  $\Phi$  being a positive function of  $\varphi$  itself obviously a positive function of  $\ell_{E,s}$ .

**Lemma 3.** All else held equal, increasing  $\sigma_0$ , sigma, and  $b_{\sigma}$  will have a negative effect on low-carbon investment.

**Proof.** This property follows from Proposition 2,  $|\varphi_r^*|$  being a negative function of the  $\sigma_{u,s}$  with a limit in zero.

# F.4. Proof of Proposition 4

**Proposition.** It is possible to define an interval  $S = [|\bar{S}; \bar{S}|]$  such that, for a given  $\rho$ ,  $b_{\ell}$  and  $\bar{\ell}$ ,  $R^* > 0 \forall S \in S$  and  $R^* \leq 0$  otherwise. S can be empty.

**Proof.** As per Proposition 1,  $u_{H,s}$  gets back closer to  $u^f$ , if the profit rate spread between high and low technologies  $\pi_{L,1} - \pi_{H,1}$  is low enough, a longer planning horizon may have a negative effect on low-carbon investment, as agents account for more time periods during which  $\pi_{L,s} - \pi_{H,s} < 0$ .

# F.5. Proof of Proposition 5

**Proposition.** The effect of a higher  $\sigma_0$  or  $\bar{\sigma}$  will depend on the sign of  $R^*$ . If  $R^* < 0$ ,  $\frac{\partial \ell_I}{\partial \sigma_0} \ge 0$  and  $\frac{\partial \ell_I}{\partial \bar{\sigma}} \ge 0$  and  $\ell_I$  is concave in  $\sigma_0$  and  $\bar{\sigma}$ . If  $R^* > 0$ ,  $\frac{\partial \ell_I}{\partial \sigma_0} \le 0$  and  $\frac{\partial \ell_I}{\partial \bar{\sigma}} \le 0$  and  $\ell_I$  is convex in  $\sigma_0$  and  $\bar{\sigma}$ . Plus, there exists an  $R' \in [R_0; R_1]$  such that  $\frac{\partial \ell_I}{\partial \Gamma}(R')$  is equal to zero.

**Proof.** Taking first the derivative of  $\ell_I$  with respect to  $\Gamma$ , we find:

$$\frac{\partial \ell_I}{\partial \Gamma} = \frac{u_1 v^* - v_1 u^*}{2\Gamma^{\frac{3}{2}} v_1^2} \tag{53}$$

With:

$$u_{1} = R_{1}\phi(\varphi_{1}) - R_{0}\phi(\varphi_{0})$$
$$u^{*} = R^{*}\phi(\varphi^{*}) - R_{0}\phi(\varphi_{0})$$
$$\phi(x) = \frac{\partial\Phi}{\partial x} \forall x$$
$$v_{1} = \Phi(\varphi_{1}) - \Phi(\varphi_{0})$$
$$v^{*} = \Phi(\varphi^{*}) - \Phi(\varphi_{0})$$

Considering the cross-derivative  $\frac{\partial \ell_I}{\partial \Gamma \partial R^*},$  we obtain:

$$\frac{\partial \mathcal{E}_I}{\partial \Gamma \partial R^*} = (v_1(\varphi^*)^2 - v_1 - \frac{u_1}{\Gamma})\frac{e^{-\frac{x^2}{2}}}{2\pi}$$

Since  $u_1 > 0$  and  $v_1 > 0$ , the determinant of  $(v_1 \varphi^* - v_1 - \frac{u_1}{\Gamma})$  is always positive. Hence, this function admits two roots in  $R^*$ ,  $R_-$  and  $R_+$ , and is strictly negative between the two corresponding local optima. Noticing that  $\frac{\partial \ell_I}{\partial \Gamma}(R_0) > 0$  and  $\frac{\partial \ell_I}{\partial \Gamma}(R_1) < 0$ , it follows that

 $\frac{\partial \ell_I}{\partial \Gamma}(R_-) > 0$  and  $\frac{\partial \ell_I}{\partial \Gamma}(R_+) < 0$ . By the theorem of intermediate value, there exists a  $R' \in [R_0; R_1]$  such that  $\frac{\partial \ell_I}{\partial \Gamma}(R')$  is equal to zero. The concavity/convexity properties of  $\ell_I$  follow from the definition of its second derivative.

# F.6. Proof of Proposition 6

**Proposition.** For a given S and  $\rho$ , there exists a  $\left(\frac{\bar{\sigma}}{\sigma_0}\right)^*$  ratio high enough such that  $\left|\frac{\partial \ell_I}{\partial \bar{\sigma}}\right| > \left|\frac{\partial \ell_I}{\partial \bar{\sigma}}\right|$ .

**Proof.** Using the result of Proposition 3, suffices to prove that, for a  $\left(\frac{\tilde{\sigma}}{\sigma_0}\right)^*$  high enough,  $\frac{\partial\Gamma}{\partial\sigma_0} > \frac{\partial\Gamma}{\partial\sigma_0}$ . Considering that the logistic sequence is well-approximated by a continuous counterpart, we write the following logistic function

 $l(x) \forall x \ge 0$ , which includes a starting value  $\sigma_0$  and a carrying capacity  $\bar{\sigma}$ :

$$l(x) = \frac{\bar{\sigma}}{1 + (\frac{\bar{\sigma}}{\sigma_0} - 1)e^{-b_{\sigma}x}}$$

Taking the corresponding derivatives:

$$\frac{\partial l}{\partial \bar{\sigma}} = \frac{1 - e^{-b_{\sigma}x}}{\left(1 + \left(\frac{\bar{\sigma}}{\sigma_0} - 1\right)e^{-b_{\sigma}x}\right)^2} \ge 0$$
$$\frac{\partial l}{\partial \sigma_0} = \frac{\left(\frac{\sigma_0}{\bar{\sigma}}\right)^2 e^{-b_{\sigma}x}}{\left(1 + \left(\frac{\bar{\sigma}}{\sigma_0} - 1\right)e^{-b_{\sigma}x}\right)^2} \ge 0$$

We consider the ratio  $\frac{\overline{\partial \sigma_0}}{\frac{\partial l}{\partial \sigma_0}}$  to determine the condition under which  $\frac{\partial l}{\partial \sigma_0} > \frac{\partial l}{\partial \overline{\sigma}}$ , which is a sufficient condition for  $\frac{\partial \Gamma}{\partial \sigma_0} > \frac{\partial \Gamma}{\partial \overline{\sigma}_0}$ .

$$\frac{\bar{\sigma}}{\sigma_0} > \sqrt{\frac{1-e^{-b_\sigma x}}{e^{-b_\sigma x}}}$$

Given that we only consider S-long logistic sequences and because  $\frac{(1-\exp(-x))}{\exp(-x)}$  is a strictly increasing function of x, a sufficient condition is that:

$$\frac{\bar{\sigma}}{\sigma_0} > \max_x \sqrt{\frac{1 - e^{-b_\sigma x}}{e^{-b_\sigma x}}} = \sqrt{\frac{1 - e^{-b_\sigma S}}{e^{-b_\sigma S}}}$$
(54)

Given Proposition 3, this defines a  $\frac{\bar{\sigma}}{\sigma_0}$  high enough for  $\left|\frac{\partial \ell_I}{\partial \bar{\sigma}}\right| > \left|\frac{\partial \ell_I}{\partial \bar{\sigma}}\right|$ .

Let us fix  $\sigma_0$  to its benchmark value of 0.01. The value of the threshold for S = 30 is around 66, which implies that the condition would hold for this planning horizon for  $\bar{\sigma} \ge 0.6$ ; that is, for a sizeable range of our parameter constellation. The condition would hold for  $\bar{\sigma} \ge 1$  with a planning horizon equal to 32, and  $\bar{\sigma} \ge 1.5$  for a planning horizon of 35. Fig. A9 below gives the range of  $\bar{\sigma}$  values for which the condition holds true for each length of the planning horizon.  $\Box$ 

# F.7. Proof of Proposition 7

**Proposition.** As belief heterogeneity approaches infinity,  $\ell_I$  will tend towards a finite value  $\tilde{\ell}_I = \frac{R^* - R_0}{R_1 - R_0}$ , where  $R^*$ ,  $R_0$  and  $R_1$  are the numerators of  $\varphi^*$ ,  $\varphi_0$  and  $\varphi_1$  respectively. We call  $\ell_I$  'full dispersion' equilibrium and it is a function of  $b_\ell$ ,  $\bar{b}$  for a given S and  $\rho$ .

**Proof.** Using l'Hôpital's Rule, we have:

$$\lim_{\Gamma \to +\infty} \mathscr{C}_I = \lim_{\Gamma \to +\infty} \frac{\Phi(\varphi^*) - \Phi(\varphi_0)}{\Phi(\varphi_1) - \Phi(\varphi_0)} = \lim_{\Gamma \to +\infty} \frac{\frac{\partial \Phi(\varphi^*) - \Phi(\varphi_0)}{\partial \Gamma}}{\frac{\partial \Phi(\varphi_1) - \Phi(\varphi_0)}{\partial \Gamma}}$$

Differentiating, we obtain:

$$\frac{\frac{\partial \Phi(\varphi^*) - \Phi(\varphi_0)}{\partial \Gamma}}{\frac{\partial \Phi(\varphi_1) - \Phi(\varphi_0)}{\partial \Gamma}} = \frac{\frac{-R^*}{2\Gamma^{\frac{3}{2}}}\phi(\varphi^*) - \frac{-R_0}{2\Gamma^{\frac{3}{2}}}\phi(\varphi_0)}{\frac{-R_1}{2\Gamma^{\frac{3}{2}}}\phi(\varphi_1) - \frac{-R_0}{2\Gamma^{\frac{3}{2}}}\phi(\varphi_0)}$$

Where  $\phi(x) = \frac{\partial \Phi}{\partial x} \forall x$ , the probability density function of a  $\mathcal{N}(0, 1)$ . Simplifying, we get:

$$\frac{\frac{\partial \Phi(\varphi^*) - \Phi(\varphi_0)}{\partial \Gamma}}{\frac{\partial \Phi(\varphi_1) - \Phi(\varphi_0)}{\partial \Gamma}} = \frac{R^* \phi(\varphi^*) - R_0 \phi(\varphi_0)}{R_1 \phi(\varphi_1) - R_0 \phi(\varphi_0)}$$



**Fig. A9.** Range of  $\bar{\sigma}$  values for which Condition (54) holds true for each length of the planning horizon. The shaded area gives the range of acceptable value, while the red line gives the value of the threshold as defined in Equation (54).

Since  $\phi(x) = \frac{1}{2\sqrt{\pi}} \exp(-\frac{x^2}{2})$ , it is easy to see that, for any *R*,  $\lim_{\Gamma \to +\infty} \phi(\frac{R}{\Gamma}) = \lim_{x \to 0^+} \phi(x)$  for  $x = \Gamma$ , which is equal to 1. Hence, we get the results

$$\lim_{\Gamma \to +\infty} \mathscr{C}_I = \frac{R^* - R_0}{R_1 - R_0}$$

Which indeed belongs to [0, 1] for  $R_0 \le R^* \le R_1$ .

If  $\ell_I = \Phi(\varphi^*)$ , (i.e. the distribution is not censored), it is straightforward that, the limit of  $\varphi^*$  being zero, that of  $\Phi(\varphi^*)$  is  $\Phi(0) = 0.5$ .

# F.8. Proof of Proposition 8

**Proposition.** An increase in the planning horizon *S* will have a positive effect if  $R_{S-1}^* + \pi_{L,S} - \pi_{H,S} > 0$ . For  $b_{\ell}$  There exists an  $s_1 \in [|1, S|]$  sufficiently large such that this condition holds. If  $\pi_{L,l} - \pi_{H,l} > 0$ , S = 1. There also exists an  $s_2 > s_1$  such that the condition reverses if  $\pi_{L,S_2} - \pi_{H,S_2}$  is negative and low enough.  $s_2$  increases with  $\bar{\ell}$  and  $b_{\ell}$ .

**Proof.** Let us assume that agents have a sufficiently large planning horizon *S*. In all generality, from Proposition 1, it is possible to decompose  $\sum_{1}^{S} \beta^{s}(\pi_{L,s} - \pi_{H,s})$  as follows:

$$\sum_{s}^{S} (\pi_{L,s} - \pi_{H,s}) = \underbrace{\sum_{1}^{s_{1}} \beta^{s}(\pi_{L,s} - \pi_{H,s})}_{<0} + \underbrace{\sum_{s_{1}+1}^{s_{2}} \beta^{s}(\pi_{L,s} - \pi_{H,s})}_{>0} + \underbrace{\sum_{s_{2}+1}^{S} \beta^{s}(\pi_{L,s} - \pi_{H,s})}_{<0}$$

Where  $s_1$  and  $s_2$  are the thresholds derived from Proposition 1. It is obvious from this decomposition that, because from  $s_2 + 1$ , only negative terms are added, a sufficiently large *S* will yield  $\sum_{0}^{S-1} \beta^s(\pi_{L,s} - \pi_{H,s}) + \beta^s(\pi_{L,S} - \pi_{H,S})$  if  $(\pi_{L,S} - \pi_{H,S})$  sufficiently negative.

Let us now assume that the planning horizon is equal to  $s_1 \le S' \le s_2$ . From Proposition 1, increasing the planning horizon to S' + 1 will have a positive effect on  $\varphi_t^*$  if  $(\pi_{L,S'} - \pi_{H,S'})$  sufficiently high, which is true for  $b_\ell$  (or  $\bar{\ell}$ ) high enough. It follows immediately

that 
$$S_2$$
 is a positive function of  $b_\ell$  and  $\bar{\ell}$ , the term  $\underbrace{\sum_{s_1+1}^{s_2} \beta^s(\pi_{L,s} - \pi_{H,s})}_{>0}$  increasing in  $b_\ell$  and  $\bar{\ell}$ .  $\Box$ 

#### References

Acemoglu, D., Jensen, M.K., 2018. Equilibrium Analysis in the Behavioral Neoclassical Growth Model. Working Paper 25363. National Bureau of Economic Research. Andre, P., Haaland, I., Roth, C., Wohlfart, J., 2021. Narratives about the macroeconomy. CEBI Working Paper Series, 2021. 18/21. CEBI, Copenhagen. Baldwin, E., Cai, Y., Kuralbayeva, K., 2020. To build or not to build? Capital stocks and climate policy. J. Environ. Econ. Manag. 100, 102235. Barrero, J.M., 2022. The micro and macro of managerial beliefs. J. Financ. Econ. 143 (2), 640–667.

#### L. Cahen-Fourot, E. Campiglio, L. Daumas et al.

Baruya, P., 2017. International finance for coal-fired power plants. Report. IEA Clean Coal Centre, London.

Beckert, J., Bronk, R., 2018. Uncertain Futures: Imaginaries, Narratives, and Calculation in the Economy. Oxford University Press, Oxford.

Binder, C., McElroy, T.S., Sheng, X.S., 2022. The term structure of uncertainty: new evidence from survey expectations. J. Money Credit Bank. 54 (1), 39–71.

Bolton, P., Kacperczyk, M., 2020. Do investors care about carbon risk? Report National Bureau of Economic Research, Cambridge, Massachusetts.

Boyer, R., 2018. Expectations, narratives, and socio-economic regimes. In: Uncertain Futures. Imaginaries, Narratives and Calculation in the Economy (Chapter 2). Oxford University Press.

Bronk, R., 2009. The Romantic Economist: Imagination in Economics. Cambridge University Press, Cambridge.

Cafferata, A., Dávila-Fernández, M.J., Sordi, S., 2021. Seeing what can (not) be seen: confirmation bias, employment dynamics and climate change. J. Econ. Behav. Organ. 189, 567–586.

Cai, Y., Lontzek, T.S., 2018. The social cost of carbon with economic and climate risks. J. Polit. Econ. 127 (6), 2684–2734.

Campiglio, E., van der Ploeg, F., 2022. Macrofinancial risks of the transition to a low-carbon economy. Rev. Environ. Econ. Policy 16 (2), 173–195.

Campiglio, E., Dietz, S., Venmans, F., 2022. Optimal Climate Policy as If the Transition Matters. CESifo Working Paper 10139. CESifo, Munich.

Campiglio, E., Lamperti, F., Terranova, R., 2023. Believe me when I say green! Heterogeneous expectations and climate policy uncertainty. Working Paper 395. Grantham Research Institute on Climate Change and the Environment, London.

Coibion, O., Gorodnichenko, Y., Kumar, S., 2018. How do firms form their expectations? New survey evidence. Am. Econ. Rev. 108 (9), 2671–2713.

Daumas, L., 2023. Financial stability, stranded assets and the low-carbon transition – a critical review of the theoretical and applied literatures. J. Econ. Surv., joes.12551.

Dávila-Fernández, M.J., Sordi, S., 2020. Attitudes towards climate policies in a macrodynamic model of the economy. Ecol. Econ. 169, 106319.

De Grauwe, P., Macchiarelli, C., 2015. Animal spirits and credit cycles. J. Econ. Dyn. Control 59, 95–117.

Dunz, N., Naqvi, A., Monasterolo, I., 2021. Climate sentiments, transition risk, and financial stability in a stock-flow consistent model. J. Financ. Stab. 54, 100872. EC, 2021. 'Fit for 55': Delivering the EU's 2030 Climate Target on the way to climate neutrality. Communication from the Commission to the European Parliament,

the Council, the European Economic and Social Committee and the Committee of the Regions COM(2021) 550 final. European Commission, Brussels. EIA, 2020. Cost and Performance Characteristics of New Generating Technologies, Annual Energy Outlook 2020.

El Ouadghiri, I., Uctum, R., 2020. Macroeconomic expectations and time varying heterogeneity: evidence from individual survey data. Appl. Econ. 52 (23), 2443–2459. Enerdata, 2021. Global Energy Scenarios Through 2050 - an in-depth look at the future of energy.

European Commission, 2019. EU Energy in Figures 2019. European Commission.

Eurostat, 2021. Eurostat - Data Explorer.

Eurostat, 2022. Eurostat Electricity production, consumption and market overview.

Fais, B., Keppo, I., Zeyringer, M., Usher, W., Daly, H., 2016. Impact of technology uncertainty on future low-carbon pathways in the UK. Energy Strategy Rev. 13, 154–168.

Figueiredo, N.C., da Silva, P.P., 2019. The "Merit-order effect" of wind and solar power: volatility and determinants. Renew. Sustain. Energy Rev. 102, 54-62.

Fouquet, R., 2010. The slow search for solutions: lessons from historical energy transitions by sector and service. Energy Policy 38 (11), 6586-6596.

Franke, R., Westerhoff, F., 2017. Taking stock: a rigorous modelling of animal spirits in macroeconomics. J. Econ. Surv. 31 (5), 1152–1182.

Franke, R., Westerhoff, F., 2018. Taking stock: a rigorous modelling of animal spirits in macroeconomics. In: Analytical Political Economy, pp. 5–38.

Gabaix, X., 2014. A sparsity-based model of bounded rationality. Q. J. Econ. 129 (4), 1661–1710.

Galanis, G., Kollias, I., Leventidis, I., Lustenhouwer, J., et al., 2022. Generalizing heterogeneous dynamic heuristic selection. Working paper. Centre for Research in Economic Theory and Its Applications CRETA, Warwick.

Geels, F.W., 2002. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. Res. Policy 31 (8–9), 1257–1274. Geisendorf, S., 2016. The impact of personal beliefs on climate change: the "battle of perspectives" revisited. J. Evol. Econ. 26 (3), 551–580.

Geroski, P., 2000. Models of technology diffusion. Res. Policy 29 (4-5), 603-625.

Giglio, S., Maggiori, M., Stroebel, J., Tan, Z., Utkus, S., Xu, X., 2023. Four Facts About ESG Beliefs and Investor Portfolios.

Grant, J., Coffin, M., 2020. Fault Lines: How diverging oil and gas company strategies link to ed asset risk. Technical report. Carbon Tracker Initiative, London.

Groom, B., Drupp, M.A., Freeman, M.C., Nesje, F., 2022. The future, now: a review of social discounting. Ann. Rev. Res. Econ. 14, 467–491.

Grubb, M., Drummond, P., Hughes, N., 2020. The Shape and Pace of Change in the Electricity Transition: Sectoral Dynamics and Indicators of Progress.

Guilmi, C.D., Galanis, G., Proaño, C.R., 2022. A Baseline Model of Behavioral Political Cycles and Macroeconomic Fluctuations. CAMA Working Paper. Centre for Applied Macroeconomic Analysis, Crawford School of Public Policy, The Australian National University.

Hommes, C., 2021. Behavioral and experimental macroeconomics and policy analysis: a complex systems approach. J. Econ. Lit. 59 (1), 149-219.

IEA, 2019. Offshore wind outlook 2019. Report. International Energy Agency, Paris.

IEA, 2020a. Energy Technology Perspectives 2020. Report. International Energy Agency, Paris.

IEA, 2020b. Projected costs of generating electricity - 2020 edition. Report. International Energy Agency, Paris.

IEA, 2020c. Tracking Power 2020. Report. IEA, Paris.

IEA, 2020d. World Energy Investment 2020. Flagship Report. International Energy Agency, Paris.

IEA, 2021. Clean energy investment trends 2021. Report. IEA, Paris.

IEA, 2022a. Global energy and climate model. Report. International Energy Agency, Paris.

IEA, 2022b. World Energy Investment 2022. Report. International Energy Agency, Paris.

IPCC, 2022. Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change.

IRENA, 2020. Global landscape of renewable energy finance 2020. Report. IRENA, Abu-Dhabi.

Kempa, K., Moslener, U., Schenker, O., 2021. The cost of debt of renewable and non-renewable energy firms. Nat. Energy 6 (2), 135-142.

Knobloch, F., Mercure, J.-F., 2016. The behavioural aspect of green technology investments: a general positive model in the context of heterogeneous agents. Environ. Innov. Soc. Trans. 21, 39–55.

Kriegler, E., Weyant, J.P., Blanford, G.J., Krey, V., Clarke, L., Edmonds, J., Fawcett, A., Luderer, G., Riahi, K., Richels, R., 2014. The role of technology for achieving climate policy objectives: overview of the EMF 27 study on global technology and climate policy strategies. Clim. Change 123 (3), 353–367.

Krueger, P., Sautner, Z., Starks, L.T., 2020. The importance of climate risks for institutional investors. Rev. Financ. Stud. 33 (3), 1067–1111.

Löffler, K., Burandt, T., Hainsch, K., Oei, P.-Y., 2019. Modeling the low-carbon transition of the European energy system-a quantitative assessment of the stranded assets problem. Energy Strategy Rev. 26, 100422.

Mankiw, N.G., Reis, R., Wolfers, J., 2003. Disagreement about inflation expectations. NBER Macroecon. Annu. 18, 209-248.

Mercure, J.-F., 2012. FTT:Power: a global model of the power sector with induced technological change and natural resource depletion. Energy Policy 48, 799–811.
Mercure, J.-F., Pollitt, H., Bassi, A.M., Viñuales, J.E., Edwards, N.R., 2016. Modelling complex systems of heterogeneous agents to better design sustainability transitions policy. Glob. Environ. Change 37, 102–115.

Nemet, G.F., Jakob, M., Steckel, J.C., Edenhofer, O., 2017. Addressing policy credibility problems for low-carbon investment. Glob. Environ. Change 42, 47–57.

Nerini, F.F., Keppo, I., Strachan, N., 2017. Myopic decision making in energy system decarbonisation pathways. A UK case study. Energy Strategy Rev. 17, 19–26. Nordeng, A., Kolos, M., Lin, Y., Liao, C., Rihel, A., Zelljadt, L., 2021. Refinitiv carbon market survey 2021. Carbon survey, Refinitiv, London.

Patton, A.J., Timmermann, A., 2010. Why do forecasters disagree? Lessons from the term structure of cross-sectional dispersion. J. Monet. Econ. 57 (7), 803-820.

#### L. Cahen-Fourot, E. Campiglio, L. Daumas et al.

#### Refinitiv, 2022. Refinitiv deals data.

Rozenberg, J., Vogt-Schilb, A., Hallegatte, S., 2020. Instrument choice and ed assets in the transition to clean capital. J. Environ. Econ. Manag. 100, 102183. Schelling, T.C., 1960. The Strategy of Conflict. Harvard University Press, Cambridge, Massachusetts.

Souder, D., Reilly, G., Bromiley, P., Mitchell, S., 2016. A behavioral understanding of investment horizon and firm performance. Organ. Sci. 27 (5), 1202-1218.

Souder, D., Badwaik, D., Bromiley, P., Mitchell, S., 2021. Measurement of long-term orientation: distinguishing between firm and country levels of analysis. In: Academy of Management Proceedings, vol. 2021. Academy of Management Briarcliff Manor, NY 10510, p. 14118.

Spiro, D., 2014. Resource prices and planning horizons. J. Econ. Dyn. Control 48, 159–175.

Steffen, B., 2020. Estimating the cost of capital for renewable energy projects. Energy Econ. 88, 104783.

Tsiropoulos, I., Nijs, W., Tarvydas, D., Ruiz Castello, P., 2021. Towards net-zero emissions in the EU energy system by 2050. JRC Technical Report. Joint Research Center, Brussels.

Van den Bremer, T.S., Van der Ploeg, F., 2021. The risk-adjusted carbon price. Am. Econ. Rev. 111 (9), 2782-2810.

van der Ploeg, F., Rezai, A., 2020. Stranded assets in the transition to a carbon-free economy. Ann. Rev. Res. Econ. 12, 281-298.

Vogt-Schilb, A., Meunier, G., Hallegatte, S., 2018. When starting with the most expensive option makes sense: optimal timing, cost and sectoral allocation of abatement investment. J. Environ. Econ. Manag. 88, 210–233.

Ware, R., Lad, F., 2003. Approximating the distribution for sums of products of normal variables. University of Canterbury, England. Tech. Rep. UCDMS 15. Xiong, W., Yan, H., 2010. Heterogeneous expectations and bond markets. Rev. Financ. Stud. 23 (4), 1433–1466.

Zeppini, P., 2015. A discrete choice model of transitions to sustainable technologies. J. Econ. Behav. Organ. 112, 187–203.